

# Accounting for U.S. post-war economic growth

del Río, Fernando and Lores, Francisco-Xavier

Universidade de Santiago de Compostela

 $27~\mathrm{May}~2020$ 

Online at https://mpra.ub.uni-muenchen.de/103394/ MPRA Paper No. 103394, posted 09 Oct 2020 11:20 UTC

# ACCOUNTING FOR U.S. POST-WAR ECONOMIC GROWTH

### By Fernando del Río D and Francisco-Xavier Lores\*

University of Santiago de Compostela

We develope a growth accounting method using the whole neoclassical growth model. We obtain three primary findings from our analysis of the U.S. economy during 1954–2017. First, the efficiency wedges in the entire period accurately account for the evolution of U.S. productivity and labor share. Second, the labor wedge was the main force driving the recovery of output and worked hours per capita in the eighties and nineties as well as after the Great Recession. Finally, if factor shares adjust competitively, the main force driving the U.S growth slowdown of both the seventies and the first decade of this century was the capital-efficiency wedge and the forces driving the U.S. Great Recession are not very different from the forces working in other OECD economies and those driving the 1982 Recession in the United States.

**Keywords:** Growth Accounting, Capital-Efficiency Wedge, Labor-Efficiency Wedge, Labor Wedge, Investment Wedge, Resource Constraint Wedge, Productivity, Labor Share, Worked Hours.

JEL classification: E1, E3, O4.

<sup>\*</sup>We thank Xunta de Galicia for financial support (GPC) GI-2060. Postal Address: Facultade de Ciencias Económicas e Empresariais Avda. do Burgo, s/n, 15782 Santiago de Compostela, Spain. e-mail: franciscoxavier.lores@usc.es bttps://orcid.org/0000-0001-7448-5045 bttps://orcid.org/0000-0003-1597-4062

#### I. INTRODUCTION

According to the evolution of output per capita and worked hours per capita, U.S. economic growth after World War II (WWII) can be divided in five periods (see Figure 1): (i) *The Long Boom* between 1954 – 1969, a period of high growth of output per capita and when worked hours per capita remained roughly stable ; (ii) *The First Growth Slowdown* between 1969 – 1982, a period of low economic growth and decline in worked hours per capita; (iii) *The Great Moderation* between 1982 – 1999, a period in which the growth rate of output per capita was around its average, but worked hours per capita experienced strong increase; (iv) *The Second Growth Slowdown* between 1999 – 2010, starting with the dot com bubble burst and finishing with the Great Recession. It was a period of low growth of output per capita and strong decline in worked hours per capita experienced strong recovery after 2010, a period in which worked hours per capita experienced strong recovery. while growth of output per capita returned close to its average rate during 1954 – 2017.

The data raise some questions. Why did the U.S. economy experience a lasting period of high economic growth after the end of WWII? Why did growth reduce after the beginning of the seventies and after the beginning of this century? Why did worked hours per capita experience strong increase in the eighties and nineties? What did the Great Recession cause?

At least from the 1950s, economists have performed growth accounting exercises using the neoclassical growth model to guide economic theory, answering such questions. Solow (1956) calculated his famous residue and concluded that to understand sustained economic growth, we had to go beyond the accumulation of productive inputs and analyze the determinants of his productive efficiency. Prescott (1998) concluded that if we want to understand the enormous differences in per capita income across countries, we need understand why the efficiency of productive inputs is so different across countries. However, these authors confined their focus on the production function. More recently, Cole and Ohanian (2002), Chari et al. (2002, 2007), Ohanian and Raffo (2012) and Brinca et al. (2016) perform business accounting exercises using the whole neoclassical growth model. They conclude that to account for the U.S. post-WWII recessions, we need to understand the reasons behind the worsening of some economic distortions, expressed as wedges in the equilibrium conditions of the standard neoclassical growth model, and not merely the changes in productive efficiency of inputs. In particular, it is necessary to understand the changes in the labor wedge (i.e., in the relationship between the marginal productivity of labor and the marginal rate of substitution between consumption and leisure).<sup>1</sup>

Our objective is identify the main forces driving U.S. economic growth after WWII. For this purpose, following the idea proposed and developed in a stochastic general equilibrium model by Chari et al. (2002, 2007), we use the whole neoclassical growth model to calculate the wedges in the equilibrium conditions allowing to match the model and the data. <sup>2</sup> Unlike these authors, we use the deterministic neoclassical growth model with perfect foresight to compute the wedges and perform the growth accounting exercise because we primarily focus on the whole U.S macroeconomic performance after WWII, not just on recessions, despite discussing the role of wedges in the main U.S. post-WWII recessions in order to compare our results with other previous ones.

The accounting growth exercise has two steps. First, using data with the equilibrium conditions of the neoclassical growth model, it measures the wedges representing the overall distortions to the relevant equilibrium conditions of the model. Second, the measured wedge values are fed back into the neoclassical growth model, one at a time, to assess how much of the observed movements of output, labor, investment, and factor shares in the period 1954 - 2017 can be attributed to each wedge. Simulating the model for each wedge alone, we calculate the wedge-alone component of each variable due to each wedge, which reflects the contribution of each wedge to the evolution of the variable in the period 1954 - 2017. This growth accounting exercise reveals the mechanisms through which fundamental processes drive economic performance. Therefore, it is a guide for determining the most promising type of theories on the primary characteristics of U.S. post-war economic growth.

Here, similar to Cole and Ohanian (2002) and Ohanian and Raffo (2012), we use the deterministic version of the neoclassical growth model to perform our accounting exercise. However, unlike them, we compute the entire paths of the five wedges mentioned above. Using the neoclassical growth model with perfect foresight compelled us to implement

<sup>&</sup>lt;sup>1</sup>Kydland and Prescott (1982) argued that shocks on efficiency of productive inputs was the main mechanism impulsing the economic cycles. The Real Business Cycle Theory follows the approach of Frisch (1933) and Slutzky (1937) who emphasized the exogenous productivity shocks being the impulse mechanism of the economic fluctuations.

<sup>&</sup>lt;sup>2</sup>Other authors have followed in their steps: Cavalcanti (2007), Chakraborty and Otsu (2013), Chakraborty (2006), Cho and Doblas-Madrid (2013), Kersting (2008), Kobayashi and Inaba (2006), Otsu (2010), Sustek (2011), Brinca (2013, 2014), Brinca et al. (2016) and Fehrle and Huber (2020). None of the previously cited works use a VES production function; they all use CD production functions.

a new method to compute the wedges. In particular, we use the nonlinear version of the first-order conditions to compute the whole paths of the wedges consistent with the equilibrium path converging to the Balanced Growth Path (BGP).<sup>3</sup> To use the neoclassical growth model in its deterministic version implies the assumption of perfect foresight; however, more interestingly, it allows using the nonlinear version of the first-order conditions to compute the wedges and simulate the model. This is an important point because our estimations suggest that the U.S. economy was transitioning between two different balanced growth paths during the analyzed period and the transitional dynamics might be relevant for the quantitative results.<sup>4</sup>

All previous works have especified a Cobb-Douglas (CD) production function to compute the wedges. It is well-known that, under perfect competition and the CD assumption, factor shares are constant. However, the US labor share is far from to have remained constant in the period 1954 - 2017 (see Figure 2). In particular, the US labor share underwent a strong fall after the beginning of this century. Moreover, it also underwent a significant decline in the seventies. The fall in the labor share both in seventies and the beginning of this century accompanied the slowdown of output growth and the downsizing of worked hours. Under the CD, movements in the factor shares must be driven by market frictions or non-competitive forces implying that rental prices move away from their marginal productivities because output elasticities for factors are constant. However, other production functions with variable ouput elasticities for factors allow for the competitive adjustment of the labor share. As discussed below, considering that changes in the labor share are driven by competitive forces modifying output elasticities for factors or by market frictions or non-competitive forces implying that rental prices depart from their marginal productivities is key to compute the labor and investment wedges and to asses their impact on output, investment and labor.

Here, we compute the wedges and simulate the model using a Cobb-Douglas (CD) production function and a Variable Elasticity of Substitution (VES) production function. Both functions belong to the family of production functions proposed by Kadiyala (1972),

<sup>&</sup>lt;sup>3</sup>To compute the wedges, we use annual data instead of quarter data, which are more suitable for the deterministic model and the implicit assumption of perfect foresight.

<sup>&</sup>lt;sup>4</sup>Chari et al. (2002, 2007) use the log-linearized versions of the first order conditions of the stochastic neoclassical growth model to simulate the model and develop a consistent method to compute the wedges. Brinca et al. (2016) develop the stochastic model and compare it with the deterministic model; however, in the deterministic model, they also use a different method than proposed in this study. In particular, they set the initial value of the investment wedge to be 1. This different method is relevant in computing the investment wedge.

which also includes the CES production function. The VES production function allows to account for the post-WWII evolution of the U.S. labor share because output elasticities for factors are not constant. Under a VES specification of the production function, five are the computed wedges as follows: capital-efficiency wedge, labor-efficiency wedge, investment wedge, labor wedge, and the resource constraint wedge. The capital-efficiency wedge and the labor-efficiency wedge are the gaps between capital and hours engaged in production, and given a factor income distribution, optimal capital and optimal labor needed to reach a certain level of output. The labor wedge reflects the gap between the marginal rate of substitution between consumption and leisure and the marginal product of labor. The investment wedge reflects the gap between the intertemporal marginal rate of substitution and the return to capital. The resource constraint wedge reflects the difference between output and its allocation to consumption and investment. Under a CD specification of the production function, the output elasticity for the productive factors is constant, and we can only compute four wedges. In particular, the labor-efficiency wedge and the capitalefficiency wedge are undetermined and only an efficiency wedge reflecting Total Factor Productivity (TFP) can be computed which is a geometric average of the labor-efficiency and the capital-efficiency wedge.

Departing from the CD hypothesis, in addition to allowing us to account for the competitive evolution of U.S. factor income shares, has significant consequences on the calculation of wedges. If we assume a CD production function, changes in the labor share are imputed to the labor and investment wedges. In particular, a fall in the labor share will be computed as a decrease in the labor wedge, and an increase in the investment wedge. In addition, leaving the CD assumption allows us to calculate an efficiency wedge for each productive factor, while using a CD specification of the production function, we can only compute an efficiency wedge reflecting TFP. The change rate of TFP equals the weighted sum of the change rates of the efficiency wedges of the productive factors, being the weights the respective output elasticities for each factor. Therefore, if the capitalefficiency wedge and the labor efficiency wedge are moving in opposite directions, a small change rate of the TFP might be computed even if both efficiency wedges are undergoing large changes. Moreover, under the CD assumption, the TFP is calculated assuming that the output elasticities for both factors are constants; however, under the VES assumption, the efficiency wedges are computed assuming that output elasticity for labor is equal to the labor share. Therefore, if capital intensity is growing, then a fall in the labor share results in lower computed growth of the TFP under the VES assumption than under the CD assumption. This implies that if the labor share was declining because the output elasticity for labor is falling (which should be the case if the marginal returns of the productive factors equal their rental prices), then a CD specification of the production function would overestimate the fall in the labor wedge and would underestimate the fall in both the investment and efficiency wedges. It is important take this into account, because of the strong fall in the U.S. labor share from the beginning of this century, and particularly, in the Great Recession. It must also be pointed out that according our method both labor and investment wedges do not depend on the particular especified production function if ouput elasticities for factors are variable and then factor shares can adjust competitively. That is to say, any other production function different of the CD production function yields the same labor and investment wedges than our VES production function. However, the efficiency wedges depend on the particular production function function.

Our results show that the efficiency wedges accurately account for the evolution of productivity and labor share of the U.S. economy in the period 1954 - 2017. Therefore, we can conclude that our accounting exercise does not call into question the validity of the usual focus of the growth literature on understanding and explaining the efficiency of productive inputs, as the engines of productivity growth. The U.S worked hours per capita did not display any significant long-run trend between 1954 and 2017, which can be accounted, because two forces worked in opposite directions. On one hand, the fall in the resource constraint wedge pushed worked hours per capita down. On the other hand, the increase in the labor wedge pushed worked hours per capita up. Moreover, our results show that the fast economic growth of the fifties and sixties was driven by the increase in the efficiency wedge (in particular, the labor-efficiency wedge if a VES production function is specified). If a VES production function is specified, the main force driving the growth slowdown and the fall in worked hours in the seventies and the first decade of this century was the decrease in the capital-efficiency wedge. However, if a CD production function is specified, the main force driving the growth slowdown and the fall in worked hours in the first decade of this century was the labor wedge, but the efficiency wedge was the main force driving the growth slowdown and the fall in worked hours in the seventies. If the First Growth Slowdown in the seventies is compared with the Second Growth Slowdown in this century, we observe the same fact: the VES specification of the production function tends to reduce the importance of the labor wedge. This is because in both periods, the labor share experiences a significant decrease. Under both specifications of the production

function, the following result is clear: the labor wedge was the main force driving the economic recovery in the eighties and the boom of worked hours in the nineties as well as the economic recovery of the Great Recession. Finally, we find that the investment wedge played a secondary role in accounting for the evolution of the main U.S. macroeconomic magnitudes in the period 1954 - 2017. Notably, the increase in the investment wedge contributed to mitigate the economic crisis of the seventies; however, its decline also contributed to slow the subsequent recovery in the eighties.

Therefore, our results suggest that to understand the Long Boom, we need to analyze the economic forces that provoked the strong increase in the labor efficiency wedge in the fifties and sixties. To understand the long periods of growth slowdown and decrease in the use of the labor factor in the seventies and in this century, we must focus on the circumstances influencing the capital-efficiency wedge and the labor wedge. However, to understand the periods of economic recovery requires focusing on the circumstances influencing the increase in the labor wedge.

The remainder of this paper is organized as follows. Section II puts our results in context. Section III describes the model, which is also calibrated. The wedges are computed in Section IV. In Section V, the model is simulated to assess the contribution of each wedge to the evolution of output, worked hours, and investment in the United States in the period 1954 - 2017. We discuss the implications of the specification of the production function in measuring the wedges in Section VI. Section VII analyzes the role of the wedges in the 82' recession and in the Great Recession. Section VIII analyzes the role of different wedges in the OECD countries. Finally, Section IX concludes.

#### II. RESULTS IN CONTEXT

First, our study relates to the plentiful literature that computes the famous Solow residual after the pioneering work by Solow (1957). It also relates to other studies that have calculated and analyzed other wedges (Chari et al., 2007, review this literature). However, in line with Chari et al. (2002, 2007) and Brinca et al. (2016), we do not confine ourselves to computing the efficiency wedge or any other wedge, but a set of wedges, which allows us to reconcile the model and the data. However, these authors focus on economic recessions, while we adopt a broader perspective, examining the U.S. post-WWII economic growth.<sup>5</sup> Our study also relates to del Río and Lores (2019) in which, using a VES production

<sup>&</sup>lt;sup>5</sup>Brinca et al. (2020) provide an excellent survey of the Business Cycle Accounting literature.

function, the authors show that the evolution of capital efficiency can accurately account for the evolution of the U.S. labor share after WWII.

To relate our study to previous ones (Chari et al., 2007, Brinca et al., 2016, and Ohanian and Raffo 2012), we have assessed the contribution of different computed wedges to the evolution of output, investment, and labor during the two main recessions suffered by the U.S. economy in the post-war period: the recession at the end of the seventies and the beginning of the eighties, which Brinca et al. (2016) call the 1982 Recession and the 2008 crisis usually called the *Great Recession*.<sup>6</sup>

Chari et al. (2007) find that the efficiency and labor wedges play primary roles in accounting for the evolution of output, labor, and investment in the 1982 Recession, and the investment wedge has essentially no role. Ohanian and Raffo (2012) find that the labor wedge accounts for almost all the fall in U.S. output per capita in the Great Recession while the efficiency wedge explains only about 20% of its drop and almost none of the drop in hours. Brinca et al (2016) find a similar result; in the U.S. Great Recession, the labor wedge had a predominant role, while the investment wedge also had an important, but secondary role. However, the efficiency wedge is not important. When using the CD specification of the production function, we also find results that although not alike in detail to those obtained by Ohanian and Raffo (2012) and Brinca et al. (2016) go along the same lines: the efficiency wedge played a prominent role in the U.S. Great Recession.

However, we obtain different findings assuming a VES production function and allowing that the the factor shares adjust competitively. For the Great Recession, we find that the strong decline in the capital-efficiency wedge was the main force driving the drop in output, investment, and worked hours, while the investment and labor wedges played significant, but secondary roles. For the 1982 Recession, the capital efficiency wedge also played a prominent role, while the labor wedge had a significant, but secondary role. Finally, the investment wedge played a negligible role. Therefore, the VES specification of the production function contributes to homogenize the economic performance of the United States in both recessions. We find that the increase in the labor wedge was the main force driving the recovery of worked hours, investment, and output, after both recessions. This is similar to Kersting (2008), who finds that the improvement in the labor wedge was necessary for the recovery of the UK economy, starting in 1984.

 $<sup>^6\</sup>mathrm{Cole}$  and Ohanian (2002) and Chari et al. (2002) focus on the Great Depression, as do Chari et al. (2007).

Ohanian and Raffo (2012) and Brinca et al. (2016) find similar results although the first ones use a model with perfect foresight and the second ones use a stochastic model. As we said above, we use a model with perfect foresight which means that agents anticipate the evolution of future wedges perfectly and thus react in the current period to actual future evolution of the wedges. Therefore, behavior of economic agents can differ significatively between both stochastic and deterministic frameworks which might explain that both models give rise to different results. Brinca et al. (2016) compared the results of both models for the United States. They find that the model with perfect foresight greatly increases the importance of the labor wedge, while reduces the importance of the investment wedge and the efficiency wedge. Therefore, in order to relate our results with that of Brinca et al. (2016), they should be interpreted in the following way: even with perfect foresight, if the labor share adjusts competitively, then according with our model the main force driving both the U.S 1982 Recession and the U.S Great Recession was the capital-efficiency wedge.<sup>7</sup>.

Brinca et al. (2016) apply the business accounting methodology to the OECD countries and find that the United States is the only OECD country in which the labor wedge played a dominant role in the Great Recession. Ohanian and Raffo (2012) also find that the drop in the labor wedge, and its contribution to the decrease in growth and worked hours during the Great Recession was much higher in the United States than in other advanced economies (see also Ohanian, 2010). These authors assume a CD production function in their works and, consequently, that the adjustment in the factor shares is driven by market frictions or non-competitive forces. However, our findings suggest that if we replace the CD assumption by a production function allowing that the factor shares adjust competitively, the forces driving the U.S. Great Recession might not be very different from the working forces in other OECD economies, and from the forces driving the U.S. 1982 Recession.

#### III. THE MODEL

A perfectly competitive representative firm produces output,  $Y_t$ , according to a VES production function and using as production factors, capital,  $K_t$ , and labor,  $H_t = h_t L_t$ , where

 $<sup>^{7}</sup>$ In any case, we are using annual data, while Brinca et al. (2016) carried out their analysis using quarterly data. It is to be expected that with annual data the results of the stochastic and deterministic models do not differ so much

 $h_t$  is worked time per worker and  $L_t$  the number of workers (which equals population),

$$Y_{t} = \left(\alpha \left(q_{t}K_{t}\right)^{\omega\psi} \left(z_{t} \left(1+\gamma\right)^{t} H_{t}\right)^{(1-\omega)\psi} + (1-\alpha) \left(z_{t} \left(1+\gamma\right)^{t} H_{t}\right)^{\psi}\right)^{\frac{1}{\psi}}$$

where  $\psi \leq 1, 0 < \alpha < 1, 0 < \omega < 1, \gamma \geq 0$  is the rate of labor-augmenting technological progress,  $q_t$  is the *capital-efficiency wedge* and  $z_t$  is the *labor-efficiency wedge*. Therefore, detrended output per worker,  $y_t = \frac{Y_t}{(1+\gamma)^t L_t}$ , is given by

$$y_t = \left[\alpha \left(q_t k_t\right)^{\omega \psi} \left(z_t h_t\right)^{(1-\omega)\psi} + (1-\alpha) \left(z_t h_t\right)^{\psi}\right]^{\frac{1}{\psi}},\tag{1}$$

where  $k_t = \frac{K_t}{(1+\gamma)^t L_t}$  is detrended capital per capita. The representative firm hires capital and labor to equalize their marginal productivities to their rental prices  $(r_t \text{ and } W_t)$ ,

$$s_t = r_t \frac{k_t}{y_t} \equiv s_{k,t} \tag{2}$$

and

$$1 - s_t = w_t \frac{h_t}{y_t} \equiv s_{h,t},\tag{3}$$

where  $w_t = \frac{W_t}{(1+\gamma)^t}$  is detrended wage per worked hour and

$$s_t = \alpha \omega \left( q_t \frac{k_t}{y_t} \right)^{\omega \psi} \left( z_t \frac{h}{y_t} \right)^{(1-\omega)\psi} \tag{4}$$

is output elasticity for capital and  $1 - s_t$  is output elasticity for labor. According to the first order conditions (2) and (3), the capital share,  $s_{k,t}$ , equals output elasticity for capital and the labor share,  $s_{h,t}$ , equals the output elasticity for labor. Moreover, both factor shares add 1.

The resource constraint is  $Y_t (1 - g_t) = C_t + X_t$ , where  $C_t$  is household consumption,  $X_t$  is investment, and  $g_t$  is the resource constraint wedge, which is the fraction of output not allocated to investment or consumption. The resource constraint can be rewritten in terms of the detrended variables per capita as follows:

$$c_t + x_t = (1 - g_t) y_t, (5)$$

where  $c_t = \frac{C_t}{(1+\gamma)^t L_t}$  is detrended consumption per capita and  $x_t = \frac{X_t}{(1+\gamma)^t L_t}$  is detrended

investment per capita.

Capital evolves according to the following move law, which includes quadratic investment adjustment costs,

$$K_{t+1} = X_t + (1 - \delta_t) K_t - \frac{\phi}{2} \left(\frac{X_t}{K_t} - \kappa\right)^2 K_t$$

where  $\phi > 0$ ,  $\kappa > 0$  and  $0 < \delta_t < 1$  is the economic depreciation rate of capital at time t. The move law of capital can be rewritten in terms of the detrended variables per capita as follows:

$$(1 + \eta_{t+1}) (1 + \gamma) k_{t+1} = x_t + (1 - \delta_t) k_t - \frac{\phi}{2} \left(\frac{x_t}{k_t} - \kappa\right)^2 k_t$$
(6)

where  $\eta_{t+1}$  is the population growth rate between t and t+1,  $\frac{L_{t+1}}{L_t} = 1 + \eta_{t+1}$ .

The representative household at time t is composed of  $L_t$  members. Each member of the representative household is endowed with one unit of time, which can be shared between leisure and labor in return for a wage,  $W_t$ . Therefore, at equilibrium,  $H_t = L_t h_t$ , where  $1 < h_t < 0$  is time offered in the labor market by a member of the representative household. The intertemporal utility function of the representative household is

$$U_t = \sum_{t=0}^{\infty} L_t \beta^t \left[ \log C_{L,t} + \mu \log \left( 1 - h_t \right) \right]$$

where  $1 - h_t$  is leisure per capita,  $0 < \beta < 1$  is the discount factor,  $C_{L,t} = \frac{C_t}{L_t}$  is consumption per capita and  $\mu > 0$  is the value of leisure relative to consumption. The household budget constraint is

$$N_t C_{L,t} + \pi_{x,t} X_t = \pi_{h,t} W_t h_t L_t + r_t K_t + B_t$$

where  $B_t$  are lump-sum transfers,  $\pi_{h,t}$  is the *labor wedge* and  $\pi_{x,t}^{-1}$  is the *investment wedge*.

The first-order conditions characterizing a maximum of the household problem are

$$\frac{1}{\beta} (1+\gamma) \frac{c_{t+1}}{c_t} = 1 + i_{t+1} \tag{7}$$

$$r_{t+1} = \frac{\pi_{x,t} \left(1 + i_{t+1}\right)}{1 - \phi\left(\frac{x_t}{k_t} - \kappa\right)} + \frac{\pi_{x,t+1}}{1 - \phi\left(\frac{x_{t+1}}{k_{t+1}} - \kappa\right)} \cdot \left[\frac{\phi}{2} \left(\frac{x_{t+1}}{k_{t+1}} - \kappa\right)^2 - \phi\left(\frac{x_{t+1}}{k_{t+1}} - \kappa\right) \frac{x_{t+1}}{k_{t+1}} - (1 - \delta_{t+1})\right]$$
(8)

$$\mu \frac{c_t}{1 - h_t} = \pi_{h,t} w_t \tag{9}$$

Here  $i_{t+1}$  is the interest rate at time t + 1. Equation (7) is the Euler equation, according to which the marginal rate of intertemporal substitution equals the gross interest rate. Equation (8) establishes that the rental price of capital equals its user cost which, in addition to the interest rate and the economic depreciation rate, also includes the investment wedge and the investment adjustment costs. Equation (9) states that the marginal rate of substitution between consumption and leisure equals the wage adjusted by the labor wedge.

Given the seven exogenous variables  $\{q_t, z_t, g_t, \pi_{x,t}, \pi_{h,t}, \delta_t, \eta_{t+1}\}_{t=0}^{\infty}$ , equation system (1)-(9) together with the transversality condition and an initial condition for the detrended capital per capita,  $k_0$ , characterize the dynamic behavior of the economy.

#### III.1. Calibrating the Balanced Growth Path (BGP)

Along a BGP, both the population growth rate and the economic depreciation rate are constants,  $\eta_t = \eta$  and  $\delta_t = \delta$ , and there are not adjustment costs,  $\frac{x_t}{k_t} = \kappa$ . Moreover, along a BGP, the resource constraint wedge,  $g_t$ , the investment wedge,  $\pi_{x,t}$ , the labor wedge,  $\pi_{h,t}$ , the capital-efficiency wedge,  $q_t$ , and the labor-efficiency wedge,  $z_t$ , remain constant as well as  $c_t$ ,  $x_t$ ,  $k_t$ ,  $h_t$ ,  $w_t$ ,  $y_t$  and  $i_t$ .<sup>8</sup> Given q, z,  $\pi_x$ ,  $\pi_h$ , and g, the following equations characterize a BGP:

$$\frac{1+\gamma}{\beta} = 1+i \tag{10}$$

$$(1+\eta)(1+\gamma) - (1-\delta) = \frac{x}{k}$$
(11)

$$sy = \pi_x \left( i + \delta \right) k \tag{12}$$

$$(1-s)y = wh \tag{13}$$

<sup>&</sup>lt;sup>8</sup>Of course, if  $q_t$  is constant, the production function displays purely labor-augmenting labor technical progress, which is a necessary condition for the existence of a BGP (see Uzawa (1961) and Jones and Scrimgeour (2004)).

$$\mu \frac{c}{1-h} = \pi_h w \tag{14}$$

$$s = \alpha \omega \left(q\frac{k}{y}\right)^{\omega\psi} \left(z\frac{h}{y}\right)^{(1-\omega)\psi} \tag{15}$$

$$y = \left[\alpha \left(qk\right)^{\omega\psi} \left(zh\right)^{(1-\omega)\psi} + (1-\alpha) \left(zh\right)^{\psi}\right]^{\frac{1}{\psi}}$$
(16)

$$c + x = (1 - g) y \tag{17}$$

The equation (10) is the Euler equation. The equation (11) establishes that the accumulation of capital is such that the ratio of investment to capital is constant. The equations (12) and (13) are the profit-maximizing conditions of the representative firm that equalize the marginal productivities of capital and labor to their rental prices. The equation (14)establishes that the marginal relation of substitution between leisure and consumption equals the wedge-adjusted wage rate. The equation (15) gives the output elasticity for capital. The equation (16) is the production function. The equation (17) is the resource constraint.

According to our strategy, we restrict the values of the parameters of the model such that they are compatible with observations from the U.S. economy. (i) We set x/y = 0.28, which is approximately the average value of the U.S. ratio in the period 1954 - 2017 and c/y = 0.62, which is around the U.S. average ratio in the last years of the sample.<sup>9</sup> (ii) According to NIPA data, in 2017, the ratio of the current-cost depreciation of NIPA fixed assets to the current-cost net stock of NIPA fixed assets was about 6.4%. Therefore, we set  $\delta = 0.064$ . (iii) We set  $\gamma = 0.0163$  and  $\eta = 0.0118$ , which are the annual average growth rates of output per worker and population in the United States in the period 1954 - 2017. (iv) According to our calculations using NIPA data, the U.S. average annual gross labor share in the period 1954 – 2017 was about 64%. Thus, we set  $\frac{wh}{y} = 0.64$ . (v) According to our calculations, the U.S. average annual worked hours in the period 1954 - 2017represented 22% of annual available time by U.S. non-institutional population between 16 and 64 years; we then set h = 0.22. (vi) We set the same annual interest rate as Brinca et al. (2016), i = 0.04. (vii) We set  $\psi = -1.9$  and  $\omega = 0.5$ , which are values near the estimated values by del Río and Lores (2019). In particular, del Río and Lores (2019) estimate  $\psi = -1.9$  and  $\omega = 0.36$ . We consider a value of  $\omega$  higher than the estimated value by del Río and Lores (2019) because to have well-defined series for  $z_t$  and  $q_t$ , it is

<sup>&</sup>lt;sup>9</sup>For  $\frac{c}{y}$ , we do take the average of the period, because this ratio has undergone a significant increases between the mid-fifties and 2017.

necessary that  $\omega > s_{k,t}$  for all t. This is because according to the proposed VES production function,  $\omega$  is an upper bound on the capital share,  $s_{k,t}$ . (viii) The relative value of leisure is normalized to 1,  $\mu = 1$ , as well as detrended output per capita, y = 1. (ix) We set  $\alpha \omega = 0.36$  so that the VES production function converges to a Cobb-Douglas production with output elasticity for capital 0.36. Therefore,  $\alpha = 0.72$ . (x) Along a BGP, the investment-capital ratio is given by (11),and hence,  $\kappa = (1 + \eta)(1 + \gamma) - (1 - \delta) = 0.092$ . The function of adjustment costs is quadratic, which is usual in macroeconomic literature and Brinca et al. (2016) use it too. To perform the quantitative analyses, we follow Brinca et al. (2016) and set  $\phi = 0.25/\kappa = 2.7174$  to obtain elasticity of the price of capital with respect to the investment-capital ratio of 0.25. Calibrated parameters, variables, and wedges are calculated by solving the equation system (10)-(17) and are displayed in Table 1.

To compare our results with others in the literature, we compare the results obtained from the simulations using our VES production function, with the results obtained from the simulations using the CD production function, arising as a limit case of the VES production function, when  $\psi$  goes to 0. In particular, the production function  $Ak^{\alpha\omega}h^{1-\alpha\omega}$ , where  $A = q^{\alpha\omega}z^{1-\alpha\omega}$  is Total Factor Productivity or the efficiency wedge. Solving the equation system (10)-(17) with the CD production function, the same variables, parameters, and wedges are calibrated than in the VES case, except that A = 1.7674. The CD specification of the production function does not allow identifying q and z, but only the Total Factor Productivity (TFP), A. In the VES case, TFP is  $A = q^s z^{1-s}$ , where s is output elasticity for capital. Under our assumptions,  $s = s_k$ , and in the VES case, we calculate TFP, as  $A = q^{s_k} z^{1-s_k}$ 

#### IV. THE WEDGES

In this section, we compute the wedges allowing to match the model and the U.S. data for the period 1954 - 2017. Our strategy is (i) compute the paths of the wedges and the path of detrended capital per capita consistent with the U.S. observations on worked hours per capita, labor share, consumption per capita, investment per capita, and output per capita in the period 1954 - 2017, assuming that the economy is converging to the previously calibrated BGP. Our strategy does not only yields wedges for the observed period but also for the future. The results are not very sensitive to the calibrated final BGP given the property of rapid convergence of the neoclassical growth model.

#### IV.1. Computing the wedges

The wedges are computed in both VES and CD cases. In the CD case, only the path of the efficiency wedge (i.e. TFP),  $A_t$ , can be identified, and not the paths of the efficiency wedges of each factor,  $q_t$  and  $z_t$ ; furthermore, there are four wedges. In the VES case, the path of the TFP is calculated as  $A_t = q_t^{s_{k,t}} z_t^{1-s_{k,t}}$ . To compute the wedges with the calibrated parameters above, we solve the equilibrium equation system given by (18)-(23) for  $k_{t+1}$ ,  $\pi_{x,t}$ ,  $\pi_{h,t}$ ,  $g_t$ ,  $q_t$ , and  $z_t$ , given the observed paths of  $\eta_{t+1}$ ,  $y_t$ ,  $c_t$ ,  $x_t$ ,  $s_{k,t}$ ,  $\delta_t$  and  $h_t$ in the period 1954 – 2017, and an initial condition for capital  $k_0$ 

$$c_t + x_t = (1 - g_t) y_t \tag{18}$$

$$(1 + \eta_{t+1}) (1 + \gamma) k_{t+1} = x_t + (1 - \delta_t) k_t$$
(19)

$$\alpha \left(q_t k_t\right)^{\omega \psi} \left(z_t h_t\right)^{(1-\omega)\psi} + \left(1-\alpha\right) \left(z_t h_t\right)^{\psi} = y_t^{\psi} \tag{20}$$

$$s_{k,t} = \alpha \omega \left( q_t \frac{k_t}{y_t} \right)^{\omega \psi} \left( z_t \frac{h_t}{y_t} \right)^{(1-\omega)\psi}$$
(21)

$$\frac{\pi_{x,t}}{1 - \phi\left(\frac{x_t}{k_t} - \kappa\right)} \frac{1 + \gamma}{\beta} \frac{c_{t+1}}{c_t} = s_{k,t+1} \frac{y_{t+1}}{k_{t+1}} - \frac{\pi_{x,t+1}}{1 - \phi\left(\frac{x_{t+1}}{k_{t+1}} - \kappa\right)} \cdot \left[\frac{\phi}{2} \left(\frac{x_{t+1}}{k_{t+1}} - \kappa\right)^2 - \phi\left(\frac{x_{t+1}}{k_{t+1}} - \kappa\right) \frac{x_{t+1}}{k_{t+1}} - (1 - \delta_{t+1})\right]$$
(22)

$$\mu c_t \frac{h_t}{1 - h_t} = \pi_{h,t} \left( 1 - s_{k,t} \right) y_t \tag{23}$$

Equation (18) is the resource constraint. Equation (19) is the capital accumulation law. Equation (20) is the production function. Equation (21) is the first order condition for capital of the representative firm. Equation (22) is the Euler condition. Finally, equation (23) is the household condition for the optimal allocation of time.

We set  $k_0 = x_0 [(1 + \eta) (1 + \gamma) - (1 - \delta_0)]^{-1}$ , where  $\gamma$  and  $\eta$  are the calibrated values,  $\delta_0 = 0.06$ , which is around the economic depreciation rate of NIPA fixed assets and durable consumer goods in the fifties, and  $x_0$  equals the U.S. investment rate in year 1954,  $x_0/y_0 = 0.2786$ , times detrended output per capita in year 1954,  $y_0 = 0.87453$ . Here, detrended output per capita at year t,  $y_t$ , is expressed relatively to the average detrended output per capita in the period 1954 - 2017 (e.g.,  $y_0 = 0.87453$  means that in year 1954, the detrended output per capita was 87.45% of the average detrended output per capita in the period 1954 – 2017), and since calibrating the BGP, we have normalized y = 1; hence, implicitly, we are assuming that the stationary level of detrended output per capita is the average value of this variable in the period 1954 – 2017. After 2017, we assume that variables  $\eta_{t+1}$ ,  $c_t$ ,  $x_t$ ,  $y_t$ ,  $s_{k,t}$ ,  $\delta_t$  and  $h_t$  follow the following process:

$$j_t = j_T e^{-\lambda(t-T)} + j \left(1 - e^{-\lambda(t-T)}\right)$$

where  $j_t$  is  $\eta_{t+1}$ ,  $c_t$ ,  $x_t$ ,  $y_t$ ,  $s_{k,t}$ ,  $\delta_t$  or  $h_t$  at period  $t \ge T$ , T = 2017 and j is the constant calibrated value above. We set  $\lambda = 0.03$ , which is around the speed of convergence estimated in most studies (see Barro and Sala-i-Martin, 1995). Our method allows us to compute converging paths of wedges from the initial period until infinity. In practice, we have computed 1000 periods.

The time-varying wedges together are presented in Figure 3. The main features of the evolution of wedges are as follows: (i) from the beginning of this century until 2009, the capital-efficiency wedge underwent a sharp decline. The decline in the capital-efficiency wedge in this century was similar to its large drop in the seventies. As a consequence of these two large drops, the capital-efficiency wedge has declined around 35% from the middle of the fifties until 2017. (ii) The labor-efficiency wedge experienced a significant increase from the mid-fifties until the mid-sixties and during the first decade of this century; however, it declined from the mid-eighties to the end of the nineties and after the Great Recession (after 2010). (iii) TFP increased from the mid-fifties to the early seventies, subsequently, it decreased until the early eighties, and remained roughly stable from the early eighties to the early years of this century. After this, the TFP decreased, the fall being higher in VES than in the CD. (iv) The labor wedge slightly decreased until the midseventies, and subsequently, experienced a strong increase until the end of the nineties. It also increased after the Great Recession (after 2010); however, it decreased from the end of the nineties, and in particular, during the Great Recession (2007 - 2010). The fall in this period is higher in CD than in VES. (v) The investment wedge remained roughly stable until the seventies, increased in the seventies, and decreased in the eighties. Its fall in the eighties is higher in VES than in CD and in VES it continued decreasing until de Great Recession, while in CDS remained roughly stable. (v) The resource constraint wedge remained roughly stable until mid-seventies and after experiencing a sustained decline from the middle of the seventies until the Great Recession, during which it increased significantly.

*Remark*. Average levels of the resource constraint wedge, the labor wedge, and the capital-efficiency wedge have undergone a significant change between the beginning of the analyzed period and its end (see Figure 3), which suggests that the BGP of the U.S. economy changed during the analyzed period. Therefore, it may be important to solve the non-linear equilibrium conditions. The economic depreciation rate of capital also experienced a significant increase during the analyzed period; however, its evolution is not presented in a figure, nor is the evolution of the population growth rate. Changes in the depreciation rate and the population growth rate provoke changes in the endogeneous variables in the same way that changes in the wedges do; hereafter, we ignore them, because their impact on the endogenous variables was negligible.

#### V. The U.S. Economic growth 1954-2017

We simulate the model to assess the extent at which the evolution of the wedges can account for the evolution of output per capita, worked hours, and investment, in the United States during 1954 - 2017. Ceteris paribus, the simulations of the model together with the corresponding observed variables (output, labor and investment) are displayed in each panel of Figures 4 to 6. Additionally, the evolution of detrended output per worked hour and the labor share, together with their wedge-alone component due to the efficiency wedge in both the CD and VES cases is displayed in Figure 7. The model is simulated for both the calibrated VES and CD production functions. The observed variable is displayed with a solid line and its simulated paths are displayed with dashedsolid lines (VES) and pointed solid lines (CD). For example, in panel (d) of Figure 4, the simulated paths of worked hours per capita (under both the VES and CD specification) are the result of simulating, given  $k_0$ , the equilibrium equation system (18)-(23), assuming that  $\pi_h$  follows the computed path above (displayed in Figure 3, panel (d)) and the remaining wedges, and both the depreciation and population growth rate remain in their steady values  $q_t = q$ ,  $z_t = z$ ,  $g_t = g$ ,  $\pi_{x,t} = \pi_x$ ,  $\eta_t = \eta$  and  $\delta_t = \delta$ . Proceeding thus, we can isolate the effect of each wedge on each variable. In particular, in the example, the effect of the labor wedge on the evolution of U.S. worked hours per capita in the period 1954 - 2017. We term the variables simulated in this way the wedge-alone components of the observed variables. The growth rates of the wedge-alone components of variables for the five subperiods in which we have divided the U.S. post-WWII growth are displayed in Table 2. In the VES case, we simulate the model jointly changing the capital-efficiency

and labor-efficiency wedges to compare the results with the results obtained in the CD case.

To summarize the results, we define the  $\sigma$ -statistic, which captures how closely a particular component tracks the changes in the underlying variable. The  $\sigma$ -statistic for detrended output per capita of wedge i is

$$\sigma_{i}^{y} = \frac{1/Var_{i}(y_{t} - y_{i,t})}{\sum_{i} (1/Var_{j}(y_{t} - y_{i,t}))}$$

where  $y_{i,t}$  is the wedge-alone component of output due to wedge *i* and  $Var_i(y_t - y_{i,t})$  is the variance of the error, (i.e., the variance of the difference between the variable and the wedge-alone component).<sup>10</sup> We compute similar statistics for worked hours per capita, detrended output per capita, and detrended investment per capita. The statistic lies in [0, 1], sums to one across the five wedges (four under the CD assumption) and reaches its maximum value of 1 when a particular output component tracks output perfectly. We compute the  $\sigma$ -statistics for the period 1954 – 2017 under the VES and CD assumptions and for the five subperiods described in the introductory section. The  $\sigma$ -statistics are displayed in Table 3. In the VES case, we compute the  $\sigma$ -statistics in the following two ways: computing the wedge-alone components due to each wedge and then computing the  $\sigma$ -statistics of each efficiency wedge and the other wedges, and computing the wedge-alone components due to both efficiency wedges together and then computing the  $\sigma$ -statistics of both efficiency wedges together and the other wedges. We do so to compare them with the CD case.

#### The whole period 1954-2017

Worked hours per capita did not display any significant increasing or decreasing trend between 1954 and 2017 (see Figure 4). In both the VES and CD cases, two forces worked in opposite directions. On the one hand, the fall in the resource constraint wedge pushed worked hours per capita down. On the other hand, the increase in the labor wedge pushed worked hours per capita up (see Figure 4, panels (d) and (f)).

In both the VES and CD cases, the efficiency wedges are the main forces driving changes in worked hours, output, and investment between 1954 and 2017. As Table 3 shows, in the VES (resp. CD) case, the  $\sigma$ -statistics for worked hours, output, and investment of

<sup>&</sup>lt;sup>10</sup>The  $\sigma$ -statistic is different from the  $\phi$  statistic proposed by Brinca et al. (2017) because while Brinca et al. (2017) use the sum of the quadratic errors to build their statistic, we use the variance of the errors. Both statistics yield the same results if the average of the variable and its wedge-alone component are equal.

both efficiency wedges are 0.43, 0.62, and 0.35 (resp. 0.35, 0.67, and 0.52). However, to accurately adjust the data, we need the other wedges, especially in accounting for the evolution of worked hours per capita and detrended investment per capita. The role played by the wedges differs along the periods in which we have divided the entire sample.

The evolution of detrended output per worked hour and the labor share, together with their wedge-alone components due to the efficiency wedges in both the CD and VES cases is displayed in Figure 7. For the effects of the wedges on the evolution of these variables, we only mention three points. First, in both the VES and CD cases, the efficiency wedges accurately account for the evolution of the U.S. output per worked hour between 1954 - 2017 (see Figure 7, panel (b)). In particular, the  $\sigma$ -statistic for productivity of the efficiency wedge in the CD case is 0.89 and the  $\sigma$ -statistic for the productivity of both efficiency wedges in the VES case is 0.83. Second, in the VES case, the efficiency wedges accurately account for the evolution of the U.S labor share between 1954 - 2017 (see Figure 7, panel (d)). In particular, the  $\sigma$ -statistic for the labor share of both efficiency wedges is 0.80. Third, in the VES case, the wedge-alone components of productivity and labor share due to the capital-efficiency wedge mostly account for the evolution of these variables, not the wedge-alone components due to the labor-efficiency wedge (see Figure 7, panel (a) and (c)). In particular, the fall of the labor share after the end of the nineties is mostly accounted for the wedge-alone component of the labor share due to the capital-efficiency wedge. In the VES case, the  $\sigma$ -statistic for productivity (resp. the labor share) of the capital-efficiency wedge is 0.36 (resp. 0.60) and the  $\sigma$ -statistic for productivity (resp. the labor share) of the labor-efficiency wedge is 0.14 (resp. 0.11).

#### The Long Boom

In both the CD and VES cases, the increase in the efficiency wedges (especially, the labor-efficiency wedge in the VES case) accurately account for the increase of detrended output per capita and detrended investment per capita in the Long Boom 1954 – 1969 (see Figure 5 and Figure 6, panels (a), (b) and (c)). As Table 3 shows, the  $\sigma$ -statistic for output (resp. investment) of the efficiency wedge in the CD case is 0.88 (resp. 0.75) and the  $\sigma$ -statistic for output (resp. investment) of both efficiency wedges in the VES case is 0.93 (resp. 0.76). In the VES case, the wedge-alone component of output (resp. investment) due to the labor-efficiency wedge performs a better adjustment of the changes in detrended output (resp. investment) per capita during the Long Boom ( $\sigma$ -statistic 0.45 for output and 0.54 for investment) than the wedge-alone component of output (resp. investment) due to the capital-efficiency wedge ( $\sigma$ -statistic 0.17 for output and 0.13 for

investment). As Table 2 shows, between 1954 and 1969, the detrended output (resp. investment) per capita grew by 26.21% (resp. 34.38%), the wedge-alone component of output (resp. investment) due to the efficiency wedge increased by 25.44% (resp. 45.75%) in the VES case (26.12% and 45.43%in the CD case).

#### The First Growth Slowdown

In both the VES and CD cases, the primarily responsibility of the subsequent growth slowdown between 1969 and 1982 lies with the decline in efficiency wedges (in particular, of the capital-efficiency wedge in the VES case) (see Figure 4, Figure 5 and Figure 6, panels (a), (b), and (c)). As Table 2 shows, under the VES assumption, in the First Growth Slowdown (between 1969 and 1982), the wedge-alone component of output due to the capital-efficiency wedge decreased by 8.67% (its  $\sigma$ -statistic is 0.42, see Table 3), and under the CD assumption, the wedge-alone component of output due to the efficiency wedge decreased by 8.13% (as Table 3 shows, its  $\sigma$ -statistic is 0.57). In the First Growth Slowdown, detrended output per capita decreased by 12.19%. In the CD case, the  $\sigma$ statistic for output of the efficiency wedge is 0.57, while, in the VES case, the  $\sigma$ -statistic for output of the capital-efficiency wedge is 0.42 and of both efficiency wedges is 0.62 (see Table 3). Detrended investment per capita experiences a significant decrease between 1969 and 1982 (it decreased by 17.73%), but with significant oscillations (see Figure 6). In both the VES and CD cases, the drop in detrended investment per capita was mostly driven by the decrease in the efficiency wedges (capital-efficiency wedge in the VES case). As Table 2 shows, in the VES case, the wedge-alone component of investment due to the capital-efficiency wedge decreased by 11.60% and the wedge-alone component of investment due to the labor-efficiency wedge decreased by 0.47%, while in the CD case, the wedge-alone component of investment due to the efficiency wedge decreased by 16.24%. In the VES case, the wedge-alone component of investment due to the laborefficiency wedge accounts for a good part of the oscillations of detrended investment per capita during the First Growth Slowdown, which is reflected in the value of the  $\sigma$ -statistic for investment of the labor-efficiency wedge, at 0.22, while the  $\sigma$ -statistic for investment of the capital-efficiency wedge is 0.18. In the VES case, the  $\sigma$ -statistic for investment of both efficiency wedges is higher, at 0.33, while, in the CD case, the  $\sigma$ -statistic for investment of the efficiency wedge is 0.32 (see Table 3).

In the VES case the evolution of worked hours per capita during the First Growth Slowdown was mostly impulsed by the capital-efficiency wedge, while, in the CD case, the evolution of worked hours per capita was mostly driven by the labor wedge, even if the efficiency wedge also played a significant role. As Table 2 shows, in the VES case, the decrease in the wedge-alone component of worked hours due to the capital-efficiency wedge between 1969 and 1982 was higher than the decrease in the wedge-alone component due to the labor wedge ( 3.14% and 1.64%, respectively). However, in the CD case, the decrease of the wedge-alone components of worked hours due to the efficiency wedge and due to the labor wedge was more similar (3.23% and 4.26%, respectively). In this period, worked hours per capita dropped by 8.1%. In the VES case, the  $\sigma$ -statistic for worked hours of the capital-efficiency wedge is 0.27, that of the labor-efficiency wedge is 0.19, and that of both efficiency wedges is 0.32, while that of the labor wedge is 0.29 (see Table 3). However, in the CD case, the  $\sigma$ -statistic for worked hours of the efficiency wedge is 0.19and that of the labor wedge is higher, at 0.57. The high value of the  $\sigma$ -statistic in the CD case reveals that in this case, the labor wedge played a significant role in accounting for the evolution of the worked hours per capita during the First Growth Slowdown.

In both the VES and CD cases, the increase in the investment wedge in the seventies contributed to reduce the growth slowdown and the downsizing of worked hours per capita (see Figure 4, Figure 5, and Figure 6, panel (e) and Table 2). As Table 2 shows, in the VES case (resp. CD case) the wedge-alone components of worked hours, output, and investment due to the investment wedge increased by 3.09%, 3.81%, and 11.64% (resp. 5.58%, 6.79%, and 22.76%) between 1969 and 1982.

#### The Second Growth Slowdown

In the VES case, the decline in worked hours per capita and the growth slowdown from the end of the past century until 2010 were mainly driven by the fall in the capitalefficiency wedge, while the labor wedge played a significant but secondary role (see Figure 4, Figure 5, and Figure 6, panels (a) and (d)). In the VES case, the  $\sigma$ -statistics for worked hours, output, and investment of the capital-efficiency wedge are 0.38, 0.41, and 0.42, while the same  $\sigma$ -statistics of the labor wedge are 0.24, 0.16, and 0.15 (see Table 3). As Table 2 shows, the wedge-alone components of worked hours, output and investment due to the capital-efficiency wedge decreased by 6.45%, 16%, and 23.32% between 1999 and 2010, while the worked hours per capita, detrended output per capita, and detrended investment per capita fell by 13.72%, 14.12%, and 31.51%.

However, in the CD case, in the Second Growth Slowdown (between 1999 and 2010), the main role corresponded to the labor wedge, even if the efficiency wedge also played a prominent role, especially, in accounting for the evolution of detrended output per capita and detrended investment per capita (see Figure 4, Figure 5 and Figure 6, panels (c) and (d)). In the CD case, the  $\sigma$ -statistics for worked hours, output, and investment of the labor wedge are 0.85, 0.33, and 0.30, while the same  $\sigma$ -statistics of the efficiency wedge are 0.06, 0.40, and 0.28 (see Table 3). As Table 2 shows, the wedge-alone components of worked hours, output, and investment due to the labor (resp. efficiency) wedge decreased by 11.35%, 6.16%, and 12.94% (resp. 0.79%, 5.77%, and 7.86%) between 1999 and 2010.

During the Second Growth Slowdown and specially in the VES case, the investment wedge strengthened the fall of worked hours per capita, detrended output per capita, and detrended investment per capita (see Figure 4, Figure 5, and Figure 6, panel (e) and Table 2). As Table 2 shows, in the VES case (resp. CD case) the wedge-alone components of worked hours, output, and investment due to the investment wedge decreased by 3.25%, 4.89%, and 12.42% (resp. 1.23%, 1.67%, and 4.92%) between 1999 and 2010. However, in the VES case, the labor-efficiency wedge contributed to reduce the growth slowdown and the fall in worked hours per capita during the Second Growth Slowdown, while its impact on the decrease of output and investment growth and worked hours was negligible during the First Growth Slowdown. As Table 2 shows, in the VES case, the wedge-alone components of worked hours, output, and investment due to the labor efficiency wedge increased by -0.14%, 1.28%, and -0.47% between 1969 and 1982, and by 1.01%, 11.39%, and 20.53% between 1999 and 2010.

#### The Recoveries

In both the VES and CD cases, the labor wedge was the main force driving the recovery of growth and worked hours per capita in the eighties and nineties and after the Great Recession(see Figure 4, Figure 5, and Figure 6, panel (d)). As Table 2 shows, in the VES case (resp. CD case) the wedge-alone components of worked hours, output, and investment due to the labor wedge increased by 21.24%, 17.33%, and 23.37% (resp. 21.64%, 17.04%, and 25.55%) between 1982 and 1999 and by 7.47\%, 5.09\%, and 8.50\% (resp. 8.91%, 5.10%, and 10.08%) between 2010 and 2017. Worked hours per capita, detrended output per capita and detrended investment per capita increased by 15.16%, 8.51%, and 18.82%, between 1982 and 1999 and by 7.21%, -2.63% and 4.85%, between 2010 and 2017.

The investment wedge's fall in the eighties contributed to reduce the recovery of output per capita, investment, and worked hours in the eighties and nineties (see Figure 4, Figure 5, and Figure 6, panel (e) and Table 2). As Table 2 shows, in the VES case (resp. CD case) the wedge-alone components of worked hours, output, and investment due to the investment wedge decreased by 3.78%, 3.13%, and 13.05% (resp. 3.66%, 1.41%, and 10.14%) between 1982 and 1999.

In the VES case, the fall in the labor-efficiency wedge also contributed to reduce recovery of output, investment and labor both after the First Growth Slowdown and the Second Growth Slowdown (see Figure 4, Figure 5, and Figure 6, panel (b) and Table 2). As Table 2 shows, in the VES case the wedge-alone components of worked hours, output, and investment due to the labor-efficiency wedge decreased by 0.03%, 2.04%, and 2.06% between 1982 and 1999 and by 0.68%, 5.15%, and 11.05% between 2010 and 2017. In the CD case, recovery of output, investment and labor after the Second Growth Slowdown was reduced by the decline of the efficiency wedge (see Figure 4, Figure 5, and Figure 6, panel (b) and Table 2). As Table 2 shows, in the CD case the wedge-alone components of worked hours, output, and investment due to the efficiency wedge decreased by 1.58%, 7.56%, and 11.81% between 2010 and 2017.

#### VI. Specification of the production function and the wedges

We have computed the labor and investment wedges under the production function with variable output elasticities for factors,  $(\pi_h, \text{ and } \pi_x^{-1})$ , assuming that the output elasticities for labor and capital equal their observed income shares, and also under the CD production function, assuming that output elasticities for capital and labor are constant,  $(s_t = s \text{ and } 1-s_t = 1-s)$ . It follows from the first order conditions (22) and (23) for labor and capital that the labor and investment wedges  $(\tilde{\pi}_h \text{ and } \tilde{\pi}_x^{-1})$  computed under the CD assumption are  $\tilde{\pi}_{h,t} = \pi_{h,t} \frac{s_{h,t}}{1-s}$  and  $\tilde{\pi}_{x,t}^{-1} \simeq \pi_{x,t}^{-1} \frac{1-s_{h,t+1}}{s}$ .<sup>11</sup> Therefore, if  $s_{h,t}$  decreases, then  $\tilde{\pi}_h$  decreases relative to  $\pi_h$  while  $\tilde{\pi}_x^{-1}$  increases relative to  $\pi_x^{-1}$ .

After the end of the nineties, and in particular, during the Great Recession (between 2007 and 2010) the U.S. labor share experienced a strong decline.<sup>12</sup> Therefore, around this time, the investment and labor wedges calculated with a VES production function and with a CD production function significantly begin to differ (see Figure 3, panel (c) and panel (d)). In particular, the sharp drop in the labor share is reflected in that the computed decline in the labor (resp. investment) wedge under a CD specification is higher (resp. lower) than its computed decline under a VES specification.

If there are constant returns to scale, the growth rate of TFP is  $g_{A,t} \equiv s_t g_{q,t} + (1-s_t) g_{z,t} = g_{Y,t} - s_t g_{K,t} - (1-s_t) g_{H,t}$ , which can be computed assuming the output elasticities for labor and capital are constants ( $s_t = s$  and  $\tilde{g}_{A,t} = g_{Y,t} - sg_{K,t} - (1-s) g_{H,t}$ ) or

<sup>&</sup>lt;sup>11</sup>The relationship between  $\tilde{\pi}_x^{-1}$  and  $\pi_x^{-1}$  is not a very good approach if the labor share  $s_h$  is changing too much.

 $<sup>^{12}\</sup>mathrm{The}$  labor share also decreased in seventies, but its decline was not so sharp

assuming that the output elasticity for labor equals its observed income share  $(1-s_t = s_{h,t})$ and  $\overline{g}_{A,t} = g_{Y,t} - (1 - s_{h,t}) g_{K,t} - s_{h,t} g_{H,t}$ . First, note that if  $g_q$  and  $g_z$  are moving in opposite directions, it might compute a low value for  $g_A$ , even if  $g_q$ , and  $g_z$ , are experiencing large variations. According to our VES production function, this happened during the Second Growth Slowdown in which both efficiency wedges largely moved in opposite directions (see Figure 3, panel (a) and panel (b)). Therefore, to calculate  $g_{A,t}$  ignoring  $g_{q,t}$  and  $g_{z,t}$  might lead to the wrong conclusion that changes in the efficiency wedges of the factors are not important in accounting for movements in output and labor. Second, if the labor share,  $s_h$ , goes down and the ratio of capital to worked hours is increasing,  $g_{K,t} - g_{H,t}$ , then  $\overline{g}_{A,t}$  decreases relative to  $\widetilde{g}_{A,t}$ . As noted above, after the end of the nineties, and particularly during the Great Recession (2007 - 2010), the U.S. labor wedge experienced a strong decline. Therefore, the efficiency wedge computed using a CD production function and the efficiency wedge implied by our VES specification differ significantly from the end of the nineties. In particular, the decline in the efficiency wedge computed using a CD production function is lower than the decline in the efficiency wedge computed using a VES production function (see Figure 3).<sup>13</sup>

Therefore, the large fall in the U.S. labor share after the end of the nineties explains that after this time, the investment wedge, labor wedge, and the efficiency wedge (i.e.,  $q^{1-s_h}z^{1-s_h}$ ) computed using a VES specification of the production function, significantly differ from those computed using a CD production function.

#### VII. The U.S. Economic recessions

In this section, we focus on the role of the wedges in the two main recessions that the U.S. economy faced after WWII. The first is the recession in the late seventies, which we call the 1982 Recession and the second is the Great Recession, which began in 2007.

Our simulations for the Great Recession under the VES specification of the production function are presented in Figure 8 and those under the CD specification are presented in Figure 10. Figure 9 and Figure 11 display the simulations for the 1982 Recession under the VES case and the CD case, respectively. The observed evolution of detrended output per capita, worked hours per capita, and detrended investment per capita are presented in panel (a) of each figure and the observed evolution of detrended output per capita along

 $<sup>^{13}</sup>$ They also significantly differ in the late sixties and early seventies, because in this period, the labor share experienced a strong swing.

with those of the wedges are presented in panel (b) of each figure. In the other four panels of each figure, we display the observed evolution of four variables (detrended output per capita, worked hours per capita, detrended investment per capita, and labor share) along with their wedge-alone components. We report the evolution of all wedges, except the resource constraint wedge, g, and their corresponding wedge-alone components. We do not include q in the analysis because its effects are not very significant. In the VES case, we also report the simulated results of the economy in which both the labor and capital efficiency wedges take their computed values together and the other wedges remain constant, to compare with the results obtained under the CD assumption. We display the percentage changes in the wedges between the start-up year of the recession (2007 or 1978) and the year in which worked hours per capita reached a minimum (2010 and 1982) in Table 4. In Table 5, we compare the changes in the observed variables between 1978 and 1982 and between 2007 and 2010, together with the changes in their wedge-alone components in the same years. In Table 6, we display the  $\sigma$ -statistics of the wedges for the periods 1978 - 1982 and 2007 - 2010. The changes in both wedges and wedge-alone components are displayed for both the VES and CD cases.

#### VII.1. The Great Recession

The primary responsibility of the fall in detrended output per capita, worked VES case. hours per capita, and detrended investment per capita lies with the sharp decline in the capital-efficiency wedge (see Figure 8). Between 2007 and 2010, the capital-efficiency wedge experienced a fall of almost 20% (see Table 4), and after 2013, it recovered slightly (see Figure 8). Between 2007 and 2010, the decline in the capital-efficiency wedge was almost of the same magnitude as the drop in detrended investment per capita, and it was more than double the drop in detrended output per capita, and worked hours per capita (see Table 4). Consequently, its contribution to the fall of these variables between 2007 and 2010 exceeded 100% for detrended output per capita and it approximately 70% and 84% for worked hours per capita and detrended investment per capita, respectively(see Table 5). The ability of the capital-efficiency wedge to account for the evolution of worked hours, output, and investment between 2007 and 2010 is reflected in the high values taken by its  $\sigma$ -statistics displayed in Table 6. In particular, its  $\sigma$ -statistic for worked hours per capita is 0.48, for detrended output per capita is 0.37, and for detrended investment per capita 0.48. However, between 2007 and 2010, the labor-efficiency wedge experienced opposite behavior (see Figure 8, panel (b)) and it contributed to reduce the negative impact of the decline in the capital-efficiency wedge on output, investment, and labor. The combined effect of both efficiency wedges account for 66% of the drop in detrended output per capita, 56% of the drop in worked hours per capita, and 45% of detrended investment per capita between 2007 and 2010 (see Table 5). The combined effect of both efficiency wedges accurately account for the evolution of the labor share during the Great Recession Era (see Figure 8, panel (f)).

Between 2007 and 2010, the fall in the investment and labor wedges was of a similar magnitude, between 3%-4%, and much lower than the decline in the capital-efficiency wedge (see Table 4). The fall in these wedges played a significant but secondary role in the fall of detrended output per capita, detrended investment per capita, and worked hours per capita (see Figure 8). Their contributions to the drop in output per capita between 2007 and 2010 also were similar; about 22% for the investment wedge and 24%for the labor wedge (see Table 5). However, the impact of the decrease in the labor wedge on worked hours per capita was higher than the impact of the decrease in the investment wedge, but the reverse can be said in the case of detrended investment per capita. In particular, the fall in the labor (resp. investment) wedge accounts for around 43% (resp. 17%) of the decrease in worked hours per capita between 2007 and 2010, while the fall in the investment (resp. labor) wedge account for 22% (resp. 17%) of the decrease in investment per capita (see Table 5). After 2010, the U.S. economy recovered (particularly intense in the worked hours per capita), which between 2010 and 2013 was driven by both the investment wedge and the labor wedge, and after 2013, only by the labor wedge (see Figure 8).

CD case. The labor wedge was the main force driving the fall and recovery of worked hours per capita, detrended investment per capita, and detrended output per capita (this last continued falling after 2010, but very slightly). Between 2007 and 2010, the fall in the labor wedge was higher than double the fall in the efficiency wedge or the investment wedge (see Table 4). The ability of the labor wedge to account for the decrease in worked hours, output, and investment between 2007 and 2010 is reflected in the high values taken by its  $\sigma$ -statistics, in Table 6. In particular, its  $\sigma$ -statistic for worked hours per capita is 0.91, for detrended output per capita is 0.49 and for detrended investment per capita 0.32. Between 2007 and 2010, (i) the decline in the labor wedge accounted for around 93% of the fall in worked hours per capita, while the decline in the efficiency and investment wedges only accounted for around 10% and 15%, respectively; (ii) the decline in the labor wedge accounted for around 56% of the fall in detrended output per capita, while the decline in the efficiency and investment wedges accounted for around 44% and 13%, respectively; (iii) the decline in the labor wedge accounted for around 38% of the fall in detrended investment per capita, while the decline in the efficiency and investment wedges accounted for around 24% and 18%, respectively. (see Table 5). Therefore, in the CD case, the efficiency wedge and the investment wedge played a secondary role in accounting for the fall of output, labor and investment during the Great Recession. After 2010, the labor wedge recovered significantly, but the efficiency wedge did not, and the investment wedge recovered slightly (see Figure 8, panel (b)). Therefore, the labor wedge also was the main responsible of recovery after 2010.

#### VII.2. The 1982 Recession

The main forces driving the fall in detrended output per capita and detrended VES case. investment per capita between 1978 and 1982 were both efficiency wedges (see Figure 9). Between 1978 and 1982, the capital efficiency wedge dropped by 6.58% and the labor efficiency wedge dropped by around 8.95%, while the labor wedge decreased by 2.3%and the investment wedge grew by 1% (see Table 4). The drop in detrended output per capita and investment per capita between 1978 and 1982 was mainly due to the fall of both efficiency wedges (see Figure 9, panel (c) and panel (e)). The combined effect of both efficiency wedges account for around 95% and 74% of the drop in detrended output per capita and detrended investment per capita, respectively (see Table 5). As Table 6 shows, the  $\sigma$ -statistic for output of both efficiency wedges is 0.91 and the  $\sigma$ -statistic for investment is 0.66. The fall in the labor wedge only accounts for around 11% and 14% of the drop in detrended output per capita and detrended investment per capita, respectively, in the period 1978 - 1982 (see Table 5). However, the joint fall in both efficiency wedges just account for 47% of the drop in worked hours per capita, while the drop in the labor wedge account for 51% (see Table 5). Therefore, the drop in the labor wedge played a secondary role in the fall of detrended investment per capita and detrended output per capita; however, it played a main role in the decrease of worked hours between 1978 and 1982. As Table 3 shows, the  $\sigma$ -statistic for worked hours of the labor wedge is 0.57, while the  $\sigma$ -statistic for worked hours of both efficiency wedges is 0.23. The combined effect of both efficiency wedges accurately account for the evolution of the labor share between 1978 - 1989 (see Figure 9, panel (f)). After 1982, the worked hours per capita, detrended investment per capita, and detrended output per capita recovered.

The increase of the labor wedge primarily led the recovery (see Figure 9, panel (c), panel (d), and panel (e)). The investment wedge has not played a significant role in the 1982 recession. The only salient fact is that after 1983, it contributed to slowing the recovery of worked hours per capita, detrended investment per capita, and detrended output per capita (see Figure 9, panel (e) panel (d), and panel (e)).

Between 1978 and 1982, the fall of the efficiency wedge was about 8.2%, while CD case. the fall of the labor wedge only was 0.8%; the investment wedge grew by 2.1% (see Table 4). The fall in the efficiency wedge accounts for most of the fall in detrended output per capita, detrended investment per capita, and worked hours per capita between 1978 and 1982 (see Figure 9, panel (c), panel (d), and panel (e)). As Table 3 shows, the  $\sigma$ statistic for output of the efficiency wedge is 0.90, for investment is 0.78, and for worked hours 0.37. Moreover, as Table 5 shows, the wedge-alone component of the efficiency wedge accounts for 104% of the drop in detrended output per capita, 74% of the drop in worked hours per capita, and 86% of the drop in detrended investment per capita, while the corresponding percentages accounted by the fall in the labor wedge are 7%, 28%, and 10%. Between 1978 and 1982, the labor wedge only played a significant but secondary role in accounting for the evolution of worked hours per capita. In particular, the  $\sigma$ -statistic for worked hours per capita of the labor wedge is 0.36 (see Table 3). The recovery of these variables after 1982 is mainly driven by the increase in the labor wedge; however, after 1984, the decrease in the investment wedge slowed down the recovery (see Figure 9, panel (c), panel (d), and panel (e)).

#### Summary of findings

1. In the Great Recession, under the VES specification of the production function, the main force driving the fall in detrended output per capita, detrended investment per capita, and worked hours per capita was the decrease in the capital-efficiency wedge. The investment wedge and the labor wedge had a significant and similar, but secondary role, in the drop of detrended output per capita, detrended investment per capita, and worked hours per capita. The fall in the investment wedge had a higher contribution to the drop in detrended investment per capita, while the fall in the labor wedge had a higher contribution to the drop in the worked hours per capita and it was about the joint contribution of both (capital and labor) efficiency

wedges. The joint evolution of both efficiency wedges accurately account for the evolution of the labor share during the Great Recession.

- 2. In the Great Recession, under the CD specification of the production function, the main force driving the fall in detrended output per capita, detrended investment per capita, and worked hours per capita between 2007 and 2010 was the decrease in the labor wedge. The contribution of the fall in the efficiency wedge to the drop in detrended output per worker and detrended investment per capita was also significant, but lower. While its contribution to the fall in worked hours per capita was minor. The investment wedge also played a significant, but secondary role in the fall of these variables during the Great Recession.
- 3. In the 1982 Recession, under the VES specification, the fall in both efficiency wedges account for most of the decrease in detrended output per capita and detrended investment per capita between 1978 and 1982 and almost half the drop in worked hours per capita. The labor wedge played a prominent role in the fall in worked hours per capita; however, it played a minor role in the fall of detrended output per capita and detrended investment per capita. The contribution of the investment wedge to the fall in detrended output per capita, detrended investment per capita, and hours per capita during the 1982's Recession was not significant, except that its decrease after 1984 slowed down the recovery. The results are similar if a CD production function is specified; however, the role of the labor wedge is lower and that of the efficiency wedge is higher, which means that in the CD the labor wedge plays also a secondary, albeit significant, role in accounting for the fall in worked hours per capita between 1978 and 1982.
- 4. According to the VES specification, both recessions differ in terms of the role of the labor-efficiency wedge. In particular, in the 1982 Recession, the labor-efficiency wedge worked in the same direction as the capital-efficiency wedge and strengthened the recession; however, in the Great Recession, between 2007 and 2010, the labor-efficiency wedge worked in the opposite direction to the capital-efficiency wedge, and its increase contributed to reducing the fall in detrended output per capita, detrended investment per capita, and worked hours per capita.
- 5. Under both the VES and CD specification, the main force driving the recovery of labor, output, and investment after the trough of both recessions was the increase

in the labor wedge.

#### VIII. THE GREAT RECESSION IN THE OECD COUNTRIES

We calibrate, compute the wedges, and simulate the model for 29 OECD countries using data from the Penn World Table 9.1 (PWT 9.1). We describe the calibration below.<sup>14</sup> The method to compute the wedges is the same as that described above.

For each country, we set  $\delta$ ,  $\gamma$ ,  $\eta$ ,  $\frac{x}{y}$ ,  $\frac{c}{y}$ ,  $\frac{wh}{y}$ , and h equal to their annual averages in the sample period. We set  $\beta = 0.97721$ ,  $\mu = 1$ , y = 1, and  $\psi = -1.9$  for all countries and  $\omega = 0.5$  for most countries. We set higher values of  $\omega$  for some countries because their reported labor shares by the PWT 9.1 are very low in some years and we need that  $s_{h,t} > 1 - \omega$  for all t to have well-defined series for  $z_t$  and  $q_t$ . We set  $\alpha$  such that  $\alpha \omega = \frac{wh}{y}$ ,  $\kappa = (1 + \eta) (1 + \gamma) - (1 - \delta)$  and  $\phi = 0.25/\kappa$ . Solving the equation system (10)-(17), the following variables, parameters, and wedges are calibrated for each country: g, i, k,  $\pi_x, \pi_h, q, z, s, x, c$  and w. The calibrated parameters that differ across countries are displayed in section III (Table 3) of the Technical Appendix.

To measure the difference between the causes of recession in two countries, j and l, we compute the following statistic for output, and worked hours:

$$\widehat{\sigma}_{j,l}^m = \sum_i \left( \sigma_i^{m,j} - \sigma_i^{m,l} \right)^2,$$

which is the sum of the squared differences between the  $\sigma$ -statistics for variable m = y, hof each wedge  $i = g, \pi_x, \pi_h, A, q \& z$  in both countries j and l. A higher value of the statistics means a higher difference between the causes of the change in the variable in each country.

In Table 7, we display the percent change in the observed fall of worked hours per capita and detrended output per capita as well as their wedge-alone components (in both the VES and CD case) in the United Sates and the other G-7 (UK, Italy, Germany, France, Japan, and Canada) countries between 2007 and the year in which worked hours per capita reached a minimum in each country (2010 in the United States). With these data, we also display the averages of the percent changes for 29 OECD countries and for

<sup>&</sup>lt;sup>14</sup>The data used for the United States in our cross-country comparative are different from the data used above and the results obtained are slightly different too. We describe the used data in section IV of the Technical Appendix.

G-7 countries. The  $\sigma$ -statistics for G-7 countries are also displayed in Table 8. The  $\hat{\sigma}$ statistics for output and worked hours for G-7 countries are displayed in Table 9 and Table
10. The percent change of the observed fall of worked hours per capita and detrended
output per capita and their wedge-alone components (in both the VES and CD case) in the
OECD countries are displayed in section III of the Technical Appendix. In addition, the  $\sigma$ -statistics for all OECD countries are contained in section III of the Technical Appendix.

In the CD case, the labor wedge is the main force driving the fall of detrended output per capita and worked hours per capita in the United States, while the efficiency wedge is of secondary importance. The wedge-alone component of output due to the labor (resp. efficiency) wedge accounts for 72% (resp. 30%) of the fall in U.S. detrended output per capita between 2007 and 2010 and the wedge-alone component of worked hours due to the labor (resp. efficiency) wedge accounts for 110%. (resp. 6%) of the fall in U.S. worked hours per capita. The U.S.  $\sigma$ -statistics for worked hours of labor wedge and efficiency wedge are 0.94 and 0.02, respectively, and 0.67 and 0.17 for output.

In the CD case, the average of the OECD countries is very different from the United States; on average, the main force driving the fall in worked hours per capita and detrended output per capita was the efficiency wedge. In particular, on average, the wedgealone component of output due to the efficiency (resp. labor) wedge accounts for 89% (resp. 18%) of the fall in detrended output per capita between 2007 and 2010, and the wedge-alone component of worked hours due to the efficiency (resp. labor) wedge accounts for 68% (resp. 54%) of the fall in worked hours per capita. The average OECD  $\sigma$ -statistics for worked hours of labor wedge and efficiency wedge are 0.31 and 0.48, respectively, and 0.12 and 0.82 for output. Therefore, in the CD case, the  $\hat{\sigma}$ -statistics for worked hours and output of the United States and the average of the OECD countries are 0.545 and 0.674, respectively.

However, in the VES case, the experience of the United States is closer to the OECD average. In particular, in the VES case, the wedge-alone component of worked hours due to the labor (resp. efficiency) wedge accounts for 74% (resp. 60%) of the fall in worked hours between 2007 and 2010, while, on average, these percentages are 63% (resp. 62%). In the VES case and regarding detrended output per capita, these percentages are farthest, but closer than in the CD case. In particular, in the VES case, the wedge-alone component of output due to the labor (resp. efficiency) wedge accounts for 45% (resp. 55%) of the fall in detrended output per capita between 2007 and 2010, while, on average, these percentages are 17% (resp. 93%).

In the VES case, the U.S.  $\sigma$ -statistics for worked hours of labor wedge and both efficiency wedges together are 0.79 and 0.13, respectively, and 0.33 and 0.46 for output. The average OECD  $\sigma$ -statistics for worked hours of labor wedge and both efficiency wedges together are 0.71 and 0.17, respectively, and are 0.1 and 0.83 for output. Therefore, in the VES case, the  $\hat{\sigma}$ -statistics for worked hours and output of the United States and the average of the OECD countries are 0.208 and 0.169, respectively, lower than the  $\hat{\sigma}$ -statistics in the CD, 0.545 and 0.674.

We now focus on G-7. Table 7 shows the results of our simulations for the G-7 (USA, UK, Italy, Germany, France, Japan, and Canada). In the CD case, the United States is an outlier because it is the only country in which the wedge-alone component of output due to the labor wedge accounts for the higher percentage of the fall in detrended output per capita. However, in the VES case, for all countries, including the United States, the main force driving the fall of output is the efficiency wedge. In the VES case, the relative importance of both wedges in the United States is more similar to the G-7 average. In particular, in the CD case, in the United States, the wedge-alone component of output due to the labor wedge accounts for 72% of the fall in detrended output per capita (the average of G-7 is 18%) and the wedge-alone component of output due to the efficiency wedge accounts for 30% (the average of G-7 is 89%). These figures are closer in the VES case; in particular, 44% - 26% and 54% - 85%.

There is higher variety in the changes in worked hours per capita and its components in the CD case. In the CD case, the labor wedge was the main force driving the fall in worked hours per capita in three countries (USA, UK, and Japan) and the efficiency wedge was the main force in the other four countries. Assuming a VES production function also contributes to homogenize the experience of the G-7 countries during the Great Recession, because in the VES case, the labor wedge was the main force driving the fall in worked hours in all countries. Moreover, in the CD case, the relative effects in the United States of both efficiency and labor wedges on worked hours are far off the average relative effects. However, in the VES case, the relative effects in the United State are closer to the G-7 average relative effects (see Table 7). In particular, in the VES case, the wedge-alone component of worked hours due to the labor wedge accounted for 74% of the fall in worked hours per capita (the G-7 average is 92%) and the wedge-alone component of worked hours due to the efficiency wedge accounts for 44% (the G-7 average is 34%). These figures are more distant in the CD case; in particular, 110% – 60% and 6% – 61%.

Moreover, in the VES (resp. CD) case, the average G-7  $\sigma$ -statistics for worked hours

of labor wedge and both efficiency wedges together are 0.71 and 0.17 (resp. 0.31 and 0.48), respectively, and 0.10 and 0.83 (resp. 0.12 and 0.82) for output. If we compare these figures with the U.S.  $\sigma$ -statistics, in the VES case, the  $\hat{\sigma}$ -statistics for worked hours and output of the United States and the average of the G-7 countries are 0.10 and 0.205, respectively, which are lower than the  $\hat{\sigma}$ -statistics in the CD, at 0.622 and 0.738.

#### IX. CONCLUSION

We performed a growth accounting exercise for the United States using the deterministic version of the neoclassical growth model. In this framework, we developed a method to compute five wedges reflecting distortions in the equilibrium conditions of the model that allow matching theory and data. We apply it to measure the wedges in the U.S. economy for the period 1954 - 2017. The model is simulated to assess the contribution of the wedges to the U.S. post-war economic growth.

Here, we argued that the measure of both investment and labor wedges crucially depends on if factor shares equal output elasticities and adjust competitively or, conversely, they do not. The reason is that if output elasticities for factors are constant, then differences between factor shares and output elasticities for factors are expressed in the wedges. In particular, the computed wedges can significantly differ when labor share differ a lot of assumed output elasticity for labor. We show that this point is empirically relevant especially due to the large fall underwent by the U.S. after the end of the nineties and during the seventies.

Our main findings are as follows:

- (i) Evolution of U.S. productivity and U.S. labor share between 1954 and 2017 can be accurately accounted by the efficiency wedges. Moreover, if the factor shares adjust competitively, the evolution of the U.S. labor share and the U.S. detrended output per worked hour between 1954 and 2017 can be mainly accounted by the evolution of the capital-efficiency wedge, and in particular, the fall in the capital-efficiency wedge could be primarily responsible for the decline in the U.S. labor share in this century and in the seventies.
- (ii) U.S worked hours per capita did not display any significant long-run trend between 1954 and 2017, which according to our model, can be accounted for because two forces were working in opposite directions. On the one hand, the fall of the resource

constraint wedge pushed worked hours per capita down. On the other hand, the increase of the labor wedge pushed worked hours per capita up.

- (iii) The outstanding growth of the fifties and sixties was mainly driven by the increase of the efficiency wedges. In particular, if the factor share adjust competitively, mainly by the increase of the labor efficiency wedge.
- (iv) If a CD production function is assumed and the adjustment of factor share is driven by market frictions or non-competitive forces, then the labor wedge was the main force driving the fall of output, investment and labor during the Second Growth Slowdown in the first decade of this century. In this period, the efficiency wedge played a significant but secondary role in accounting for the decrease of output, investment and labor. However, during the First Growth Slowdown in the seventies, the main force driving the evolution of output and investment was the efficiency wedge, while the main force driving the fall of of labor was the labor wedge.
- (v) If a VES production function is assumed and the adjustment of factor shares is driven by competitive forces, the decline of output, investment and labor both during the First Growth Slowdown and during the Second Growth Slowdown was driven by the capital-efficiency wedge. In both periods, the labor wedge played a significant but secondary role in accounting for the evolution of output, investment an labor.
- (vi) It follows from (iv) and (v) that allowing that factor shares adjust competitively assuming a production function with variable output elasticities for factors helps to harmonize both periods of growth slowdown. However, there are two differences between both periods. First, the investment wedge played a significant but secondary role in accounting for the fall of output, investment and labor in the Second Growth Slowdown, but it contributed to reduce the decrease of output, investment and labor in the First Growth Slowdown. Second, the efficiency-labor wedge played a negligible role in accounting for the fall of output, investment and labor in the First Growth Slowdown, but it contributed to reduce the fall of output, investment and labor in the First Growth Slowdown.
- (vii) The labor wedge was the main force driving the recovery of growth and worked hours per capita in the eighties and nineties as well as after the Great Recession.

We have applied the methodology to understand the two main economic recessions that the American economy faced after WWII. Our primary finding is that allowing the factor share to adjust competitively or do not is crucial for identifying the causes of the recessions. Briefly, if a CD production function is specified and factor shares do not adjust competitively, the main force driving the U.S. 1982's recession was the efficiency wedge; however, the main force driving the U.S. Great Recession was the labor wedge. However, if a VES production function is specified and factor shares adjust competively, the main force driving both U.S. recessions is the capital-efficiency wedge, even if the labor wedge played a prominent role overall in the fall in worked hours per capita during the U.S. Great Recession. Moreover, if a CD production function is specified, the Great Recession in the United States appears to be different from the Great Recession in the G-7 countries, because only in the U.S. economy did the labor wedge play the main role in accounting for both the growth slowdown of output per capita and the fall in worked hours per capita. However, if a VES production function is specified, the performance of G-7 countries appears to be more homogeneous.

Finally, our analysis has the following implication: to understand the periods of economic rise and recession recovery, we must to understand the factors improving the labor wedge. Similarly, to understand the periods of economic decline and recession, we must first understand the factors worsening the capital efficiency wedge.

Regarding future extensions of this work, we must point out that we have carried out the analysis in a perfect forsight environment because we are particularly interested in the analysis of the path of U.S. economic growth. However, specially in order to delve into the causes of economic recessions, it would be interesting to see if our main findings are kept in a stochastic environment. We conjecture that this will be the case, although, in line with the findings of Brinca et al. (2016), possibly the role played by the labor wedge is reduced, which will reinforce our results, but, to confirm or refute our conjecture, it would be necessary to carry out the complete analysis of a dynamic general equilibrium model.

#### REFERENCES

- BARRO, R. J. and SALA-I-MARTIN, X. (1995). *Economic Growth*, McGraw Hill International Editions.
- BRINCA, P. (2014). Distortions in the Neoclassical Growth Model: A Cross Country Analysis. *Journal of Economic Dynamics and Control*, 47, 1-19.

- BRINCA, P. (2013). Monetary Business Cycle Accounting for Sweden. *The B.E. Journal* of *Macroeconomics*, 13(1), 2194-6116.
- BRINCA, P., CHARI, V. V., KEHOE, P. J. and MCGRATTAN, E. R. (2016). Accounting for business cycles. In *Handbook of Macroeconomics* (Vol. 2, pp. 1013-1063). Elsevier.
- BRINCA, P., COSTA-FILHO, J. and LORIA, F. (2020). Business Cycle Accounting: what have we learned so far?. *MPRA Paper No. 100180*, https://mpra.ub.uni-muenchen. de/100180/
- CAVALCANTI, T. (2007). Business cycle and level accounting: the case of Portugal, *Portuguese Economic Journal.*
- CHARI, V., KEHOE, P. J. and MCGRATTAN E. R. (2002). Accounting for the Great Depression. *American Economic Review*, 92 (2), 22-27.
- CHARI, V., KEHOE, P. J., and MCGRATTAN, E. R. (2007). Business cycle accounting. *Econometrica*, 75(3), 781-836.
- CHAKRABORTY, S. (2006). Business cycle accounting of the Indian economy. *The Indian Economic Journal*.
- CHAKRABORTY, S., and OTSU, K. (2013). Business cycle accounting of the BRIC economies. *The B.E. Journal of Macroeconomics*, 13(1), 1-33.
- CHO, D., and DOBLAS-MADRID, A. (2013). Business Cycle Accounting East and West: Asian Finance and the Investment Wedge. *Review of Economic Dynamics*, 16(4), 724-744.
- COLE, H. L. and OHANIAN, L. E. (2002). The US and UK Great Depressions through the lens of neoclassical growth theory. *American Economic Review*, 92(2), 28-32.
- DEL RÍO, F. and LORES, F. X. (2019). The decline in capital efficiency and labour share. *Economica*. https://doi.org/10.1111/ecca.12279.
- FEHRLE, D. and HUBER, J. (2020). Business cycle accounting for the German fiscal stimulus program during the Great Recession. University of Augsburg, Beitrag No. 339.

- FRISCH, R. (1933). Propagation and impulse problems in dynamic economics. *Economic* essays in honour of Gustav Cassell. London: George Allen & Unwin, 171-205.
- KADIYALA, K.R. (1972). Production functions and elasticity of substitution. Southern Economic Journal, 38(3), 281-284.
- KERSTING, E. K. (2008). The 1980s recession in the UK: A business cycle accounting perspective. *Review of Economic Dynamics*, 11 (1), 179-191.
- KOBAYASHI, K. and INABA, M. (2006). Business cycle accounting for the Japanese economy. Japan and The World Economy, 18(4), 418-440. https://doi.org/10. 1016/j.japwor.2006.04.003.
- KYDLAND, F. E. and PRESCOTT, E. C. (1982). Time to build and aggregate fluctuations. *Econometrica*, 50(6), 1345-1370.
- MCGRATTAN, E. R. and PRESCOTT, E. C. (2010). Unmeasured investment and the puzzling US boom in the 1990s. *American Economic Journal: Macroeconomics*, 2(4), 88-123.
- OHANIAN, L.E. (2010). The economic crisis from a neoclassical perspective. *Journal of Economic Perspectives*, 24(4), 45-66.
- OHANIAN, L. and RAFFO, A. (2012). Aggregate hours worked in OECD Countries: New measurement and implications for business cycles. *Journal of Monetary Economics*, 59(1), 40-56.
- OTSU, K. (2010). A neoclassical analyses of Asian crisis: business cycle accounting for a small open economy. *The B.E. Journal of Macroeconomics*, 10(1), 1-39.
- PRESCOTT, E. (1998). Needed: A theory of total factor productivity. *International Economic Review*, 39(3), 525-551.
- SLUTZKY, E. 1937. The Summation of Random Causes as the Source of Cyclic Processes. *Econometrica*, 5(2), 105-146.
- SOLOW, R. M. (1957). Technical change and the aggregate production function. *Review* of *Economics and Statistics*, 39, 312-320.

- SUSTEK, R. (2011). Monetary Business Cycle Accounting, *Review of Economic Dynamics*, 14(4), 592-612.
- UZAWA, H. (1961). Neutral inventions and the stability of growth equilibrium. *Review* of *Economic Studies*, 28(2), 117-24.

Model parame	eters obtained exogenously or esti	mated.
Parameter	Description	Value
$\eta$	Population Growth Rate	0.0118
$\gamma$	Growth Rate of Output per Worker	0.0163
$\delta$	Depreciation Rate of Capital	0.0640
$\psi$	Production Function Parameter	-1.900
$\omega$	Production Function Parameter	0.5000
lpha	Production Function Parameter	0.7200
$\mu$	Relative Value of Leisure	1.0000
Model par	ameters obtained solving the mod	del.
$\beta$	Discount Factor	0.9772
bgp variables		
$\overline{q}$	Capital Efficiency Wedge	0.3296
z	Labor Efficiency Wedge	4.5454
$\pi_x$	Investment Wedge	1.1409
$\pi_h$	Labor Wedge	0.2732
g	Resource Constraint Wedge	0.10
k	Capital-Output Ratio	3.0338
y	Output per Capita	1.0000
h	Worked Hours per Capita	0.2200
x	Investment Rate	0.2800
С	Consumption to Output Ratio	0.6200
s	Capital-Output Elasticity	0.3600
i	Interest Rate	0.0400

TABLE 1 MODEL PARAMETERS AND BGP VARIABLES

TABLE $2$	
WEDGE-ALONE COMPONENTS, GROWTH RATE	(%).

				VES	5				C	D	
	Data	q	z	q and $z$	$\pi_x$	$\pi_h$	g	A	$\pi_x$	$\pi_h$	g
					$\mathrm{Th}$	e long	boom				
h	2.21	3.79	1.76	5.09	-2.10	-2.70	-0.49	5.01	-2.90	-1.49	-1.31
У	26.21	11.85	11.85	25.44	3.71	-2.85	5.09	26.12	1.59	-1.71	4.06
$\mathbf{X}$	34.38	15.62	29.52	45.75	-6.10	-2.85	-2.51	45.43	-6.11	-1.89	-1.72
				ſ	The first	growt	h slow	down			
h	-8.10	-3.14	-0.14	-3.36	3.09	-1.64	-3.40	-3.23	5.58	-4.26	-3.48
У	-12.19	-8.67	1.28	-7.48	3.81	-1.89	-2.45	-8.13	6.79	-3.79	-2.40
х	-17.73	-11.60	-0.47	-12.34	11.64	-2.77	-5.10	-16.24	22.76	-7.56	-4.53
					The g	reat m	oderat	ion			
h	15.16	2.91	-0.03	2.92	-3.78	21.24	-4.41	1.95	-3.66	21.64	-4.55
У	8.51	3.03	-2.04	0.93	-3.13	17.33	-3.78	-0.44	-1.41	17.04	-3.76
х	18.82	11.13	-2.06	9.23	-13.05	23.37	-0.18	5.28	-10.14	25.55	-0.01
				$\mathrm{Tl}$	ne secon	d grov	wth slo	wdown			
h	-13.72	-6.45	1.01	-5.01	-3.25	-5.10	1.47	-0.79	-1.23	-11.35	1.51
у	-14.12	-16.00	11.39	-6.43	-4.89	-1.95	1.27	-5.77	-1.67	-6.16	1.30
х	-31.51	-23.32	20.53	-6.64	-12.42	-6.21	-1.56	-7.86	-4.92	-12.94	-1.61
				$\mathbf{T}$	he great	recess	sion re	covery			
h	7.21	0.14	-0.68	-1.05	1.70	7.47	-1.45	-1.58	1.27	8.91	-1.51
у	-2.63	-1.96	-5.15	-6.97	0.42	5.09	-1.24	-7.56	0.87	5.10	-1.26
x	4.85	0.07	-11.05	-11.91	7.06	8.50	1.15	-11.81	4.38	10.08	1.13

## TABLE 3

## $\sigma\text{-}\mathrm{STATISTICS}.$

	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$												
Variable	q	z	$\pi_x$	$\pi_h$	g	q and $z$	$\pi_x$	$\pi_h$	g	A	$\pi_x$	$\pi_h$	g
						Entire	e samp	le					
h	0.33	0.26	0.20	0.10	0.12	0.43	0.27	0.13	0.17	0.35	0.27	0.24	0.15
У	0.38	0.11	0.25	0.05	0.21	0.62	0.18	0.04	0.16	0.67	0.11	0.06	0.16
y/h	0.36	0.14	0.18	0.17	0.15	0.76	0.08	0.08	0.07	0.89	0.03	0.04	0.04
x	0.29	0.14	0.26	0.10	0.21	0.35	0.30	0.11	0.24	0.52	0.16	0.11	0.21
$1$ - $s_k$	0.60	0.11	0.09	0.12	0.08	0.80	0.07	0.08	0.05	0.00	0.00	0.00	0.00
						The lo	ng boo	om					
h	0.19	0.13	0.21	0.33	0.15	0.20	0.24	0.38	0.18	0.16	0.24	0.39	0.21
У	0.17	0.45	0.14	0.08	0.16	0.93	0.03	0.02	0.03	0.93	0.02	0.02	0.03
y/h	0.15	0.43	0.16	0.09	0.16	0.84	0.06	0.03	0.06	0.82	0.06	0.04	0.07
x	0.13	0.54	0.10	0.11	0.11	0.73	0.09	0.09	0.09	0.76	0.08	0.07	0.08
$1 extsf{-}s_k$	0.26	0.17	0.16	0.26	0.15	0.59	0.12	0.19	0.11	0.00	0.00	0.00	0.00
					$\mathbf{T}$	ne first gro	$\mathbf{s}$	lowdov	vn				
h	0.27	0.19	0.12	0.23	0.19	0.32	0.16	0.29	0.23	0.19	0.07	0.57	0.16
У	0.42	0.12	0.09	0.14	0.22	0.62	0.08	0.12	0.19	0.57	0.06	0.17	0.20
y/h	0.30	0.15	0.15	0.25	0.14	0.69	0.09	0.14	0.08	0.76	0.06	0.12	0.06
x	0.18	0.22	0.11	0.21	0.28	0.33	0.12	0.24	0.31	0.32	0.08	0.27	0.33
$1-s_k$	0.44	0.14	0.13	0.18	0.11	0.72	0.09	0.12	0.07	0.00	0.00	0.00	0.00
						The great	mode	ration					
h	0.16	0.10	0.05	0.61	0.07	0.17	0.06	0.69	0.08	0.17	0.08	0.66	0.10
У	0.25	0.17	0.22	0.14	0.22	0.41	0.23	0.14	0.22	0.30	0.32	0.14	0.23
y/h	0.13	0.37	0.12	0.23	0.15	0.56	0.11	0.20	0.13	0.60	0.07	0.21	0.12
x	0.12	0.26	0.18	0.13	0.31	0.32	0.19	0.15	0.34	0.38	0.21	0.12	0.30
$1-s_k$	0.27	0.27	0.16	0.17	0.13	0.51	0.17	0.18	0.14	0.00	0.00	0.00	0.00
					The	e second g	$\operatorname{rowth}$	slowdo	own				
$\mathbf{h}$	0.38	0.10	0.19	0.24	0.09	0.36	0.24	0.29	0.11	0.06	0.06	0.85	0.04
У	0.41	0.04	0.29	0.16	0.10	0.48	0.28	0.15	0.10	0.40	0.17	0.33	0.10
y/h	0.03	0.03	0.33	0.20	0.41	0.58	0.15	0.09	0.18	0.21	0.29	0.10	0.40
x	0.42	0.05	0.24	0.15	0.13	0.29	0.33	0.20	0.18	0.28	0.23	0.30	0.18
$1-s_k$	0.56	0.20	0.10	0.06	0.08	0.97	0.01	0.01	0.01	0.00	0.00	0.00	0.00
					Th	e great ree	cession	recov	ery				
$\mathbf{h}$	0.02	0.02	0.03	0.91	0.02	0.02	0.03	0.93	0.02	0.05	0.09	0.82	0.05
У	0.30	0.03	0.12	0.02	0.53	0.05	0.17	0.02	0.75	0.04	0.13	0.02	0.80
y/h	0.16	0.43	0.14	0.18	0.10	0.68	0.11	0.14	0.07	0.63	0.08	0.22	0.07
x	0.10	0.02	0.56	0.15	0.17	0.02	0.62	0.17	0.19	0.03	0.63	0.12	0.22
$1-s_k$	0.26	0.06	0.26	0.12	0.29	0.24	0.29	0.13	0.33	0.00	0.00	0.00	0.00

					VES	5				С	D			
У	h	x	q	z	q and $z$	$\pi_x$	$\pi_h$	g	A	$\pi_x$	$\pi_h$	g		
			The 1982 Recession											
-9.83	-4.28	-20.94	-6.58	-8.95	-7.57	1.09	-2.29	8.90	-8.18	2.13	-0.82	8.90		
						The	Great F	Recession	n					
-8.22	-7.71	-23.42	-19.72	9.67	-8.98	-2.70	-3.84	29.86	-2.59	-2.14	-7.13	29.86		

TABLE 4 Wedge changes, deviation from peak (hours trough).

Table 5 Wedge-alone components, deviation from peak (hours trough).

				VES					Cl	D	
	Data	q	z	q and $z$	$\pi_x$	$\pi_h$	g	A	$\pi_x$	$\pi_h$	g
					The 1	.982 R	ecessic	n			
h	-4.28	-1.17	-0.77	-1.99	0.45	-2.17	0.26	-3.17	0.65	-1.21	0.36
у	-9.83	-3.36	-6.08	-9.31	1.61	-1.10	0.14	-10.29	2.34	-0.75	0.24
$\mathbf{X}$	-20.94	-4.51	-11.30	-15.48	1.81	-2.94	-3.78	-18.09	3.93	-2.27	-3.53
					The G	Freat F	lecessi	on			
h	-7.71	-5.47	0.62	-4.35	-1.31	-3.31	1.51	-0.73	-1.16	-7.18	1.74
у	-8.22	-11.56	6.81	-5.48	-1.83	-1.99	0.97	-3.65	-1.13	-4.68	1.11
$\mathbf{x}$	-23.42	-19.79	11.66	-9.06	-5.28	-4.04	-5.28	-5.63	-4.21	-9.01	-5.00

					VE	CS					С	D	
Variable	q	z	$\pi_x$	$\pi_h$	g	q  and  z	$\pi_x$	$\pi_h$	g	A	$\pi_x$	$\pi_h$	g
						The 1982	Rece	ssion					
h	0.16	0.14	0.09	0.52	0.10	0.23	0.10	0.57	0.10	0.37	0.13	0.36	0.14
У	0.20	0.49	0.08	0.13	0.10	0.91	0.02	0.04	0.03	0.90	0.02	0.04	0.03
y/h	0.13	0.69	0.05	0.05	0.07	0.80	0.06	0.06	0.08	0.79	0.06	0.07	0.08
x	0.17	0.40	0.11	0.15	0.17	0.66	0.08	0.12	0.14	0.78	0.05	0.08	0.09
$1$ - $s_k$	0.06	0.53	0.17	0.14	0.10	0.51	0.21	0.17	0.12	0.00	0.00	0.00	0.00
						The Great	t Rece	ession					
h	0.48	0.06	0.11	0.30	0.05	0.39	0.14	0.40	0.07	0.03	0.04	0.91	0.02
У	0.37	0.04	0.23	0.26	0.10	0.61	0.15	0.17	0.06	0.27	0.16	0.49	0.08
y/h	0.03	0.04	0.29	0.35	0.29	0.46	0.17	0.21	0.17	0.21	0.32	0.19	0.28
x	0.48	0.04	0.17	0.14	0.17	0.31	0.24	0.21	0.24	0.21	0.24	0.32	0.24
$1-s_k$	0.53	0.25	0.08	0.05	0.08	0.93	0.02	0.02	0.02	0.00	0.00	0.00	0.00

Table 6  $\sigma$ -statistics for two recessions.

TABLE	7
-------	---

The Great Recession in G7 Countries. Wedge-alone components, deviation (%) from peak (hours trough).

					VE	S			С	D		
		Data	q	z	q and $z$	$\pi_x$	$\pi_h$	g	A	$\pi_x$	$\pi_h$	g
CAN	у	-11.54	3.73	-2.55	-9.83	0.95	-2.89	-1.04	-11.47	0.97	-0.19	-1.18
2009	$\mathbf{h}$	-4.74	2.30	-1.23	-0.35	0.79	-4.37	-1.15	-4.20	0.57	-0.23	-1.39
DEU	у	-10.54	10.51	-3.93	-8.25	-0.71	-1.57	-0.72	-10.43	-0.70	2.33	-0.76
2009	$\mathbf{h}$	-1.84	5.94	-1.81	1.34	-0.09	-1.96	-0.59	-3.91	-0.28	4.97	-0.87
FRA	у	-15.24	-2.31	-2.30	-15.53	2.39	-2.66	0.28	-15.56	1.66	-1.58	0.36
2016	$\mathbf{h}$	-4.25	-1.49	-0.97	-3.51	1.70	-3.07	0.28	-4.79	0.81	-0.60	0.42
GBR	У	-9.78	-1.27	-1.74	-7.44	2.61	-6.00	1.10	-7.63	2.33	-5.72	1.25
2011	$\mathbf{h}$	-4.49	-0.95	-0.79	-1.60	4.02	-8.91	2.70	-2.23	4.31	-9.10	2.41
ITA	У	-23.60	-8.80	-3.93	-22.82	2.97	-5.37	0.83	-22.49	3.32	-5.44	1.20
2014	$\mathbf{h}$	-12.64	-2.88	-1.86	-5.57	2.13	-10.19	1.05	-5.97	1.56	-9.72	1.23
JPN	у	-13.66	9.04	-6.50	-11.08	-0.58	-1.90	-0.23	-12.72	-0.65	-0.13	-0.16
2009	$\mathbf{h}$	-4.91	5.52	-3.04	-0.64	-0.02	-3.51	-0.21	-4.20	-0.24	-0.13	-0.24
USA	У	-8.46	-11.19	2.14	-4.65	-1.54	-3.80	1.58	-2.56	-1.27	-6.08	1.75
2010	$\mathbf{h}$	-9.09	-5.40	1.10	-4.06	-0.27	-6.72	2.38	-0.57	-0.20	-10.04	2.54
G7	у	-13.25	-0.04	-2.68	-11.37	0.86	-3.45	0.25	-11.83	0.80	-2.40	0.35
	$\mathbf{h}$	-5.99	0.43	-1.22	-2.05	1.18	-5.53	0.63	-3.69	0.93	-3.55	0.58
OECD	у	-15.12	-6.82	-2.28	-14.17	0.89	-2.57	0.54	-13.93	1.58	-2.77	0.86
	$\mathbf{h}$	-7.00	-3.44	-1.09	-4.37	1.11	-4.45	1.24	-4.78	0.94	-3.78	1.31

In the first column we indicate the year of the trough. The year of the peak is 2007 for all countries.

TABLE 8 The Great Recession in G7 Countries.  $\sigma\textsc{-statistics}$  for G7 Countries.

						VF	S					С	D	
Country		q	z	$\pi_x$	$\pi_h$	g	q and $z$	$\pi_x$	$\pi_h$	g	A	$\pi_x$	$\pi_h$	g
CAN	У	0.07	0.34	0.14	0.26	0.19	0.87	0.03	0.06	0.04	0.99	0.00	0.00	0.00
	$\mathbf{h}$	0.02	0.08	0.04	0.76	0.10	0.04	0.04	0.81	0.11	0.90	0.02	0.02	0.06
$\mathbf{DEU}$	У	0.02	0.72	0.08	0.10	0.08	0.98	0.01	0.01	0.00	0.83	0.06	0.04	0.06
	$\mathbf{h}$	0.00	0.02	0.01	0.96	0.01	0.00	0.01	0.98	0.01	0.20	0.54	0.06	0.20
$\mathbf{FRA}$	У	0.18	0.30	0.12	0.23	0.16	0.87	0.03	0.06	0.04	0.88	0.03	0.05	0.04
	$\mathbf{h}$	0.16	0.24	0.08	0.42	0.10	0.42	0.08	0.41	0.09	0.59	0.10	0.20	0.11
$\mathbf{GBR}$	У	0.06	0.51	0.06	0.30	0.07	0.75	0.04	0.17	0.04	0.91	0.02	0.06	0.02
	$\mathbf{h}$	0.18	0.49	0.06	0.19	0.08	0.44	0.11	0.32	0.13	0.63	0.04	0.26	0.08
ITA	У	0.23	0.31	0.10	0.24	0.12	0.99	0.00	0.00	0.00	0.99	0.00	0.00	0.00
	$\mathbf{h}$	0.06	0.06	0.03	0.81	0.03	0.11	0.03	0.82	0.03	0.24	0.06	0.64	0.06
$\mathbf{JPN}$	У	0.03	0.76	0.07	0.09	0.06	0.89	0.03	0.04	0.03	0.98	0.01	0.01	0.01
	$\mathbf{h}$	0.01	0.21	0.04	0.70	0.04	0.07	0.05	0.83	0.05	0.81	0.07	0.07	0.06
$\mathbf{USA}$	У	0.58	0.03	0.10	0.24	0.05	0.46	0.14	0.33	0.07	0.17	0.11	0.67	0.06
	$\mathbf{h}$	0.21	0.03	0.04	0.70	0.02	0.13	0.05	0.79	0.03	0.02	0.02	0.94	0.01
Avg.	У	0.17	0.43	0.10	0.21	0.10	0.83	0.04	0.10	0.03	0.82	0.03	0.12	0.03
$\mathbf{G7}$	$\mathbf{h}$	0.09	0.16	0.04	0.65	0.05	0.17	0.05	0.71	0.06	0.48	0.12	0.31	0.08
Avg.	у	0.25	0.29	0.14	0.19	0.14	0.78	0.07	0.08	0.07	0.75	0.08	0.09	0.08
OECD	h	0.20	0.24	0.13	0.35	0.08	0.32	0.17	0.40	0.11	0.37	0.19	0.32	0.11

Table 9 The Great Recession in G7 Countries.  $\hat{\sigma}$ -statistics for output in G7 countries.

					V	ES								C	D			
	CAN	DEU	FRA	GBR	ITA	JPN	USA	Avg.G7	Avg.OECD	CAN	DEU	FRA	GBR	ITA	JPN	USA	Avg.G7	Avg.OECD
CAN	0.000	0.016	0.000	0.030	0.019	0.001	0.260	0.004	0.011	0.000	0.034	0.017	0.011	0.000	0.000	1.141	0.045	0.078
DEU	0.016	0.000	0.018	0.086	0.000	0.011	0.402	0.033	0.053	0.034	0.000	0.004	0.009	0.034	0.029	0.835	0.008	0.009
$\mathbf{FRA}$	0.000	0.018	0.000	0.028	0.020	0.001	0.254	0.003	0.011	0.017	0.004	0.000	0.001	0.016	0.013	0.903	0.009	0.022
GBR	0.030	0.086	0.028	0.000	0.090	0.038	0.119	0.013	0.010	0.011	0.009	0.001	0.000	0.011	0.008	0.930	0.011	0.033
ITA	0.019	0.000	0.020	0.090	0.000	0.013	0.413	0.036	0.060	0.000	0.034	0.016	0.011	0.000	0.000	1.135	0.043	0.078
$_{\rm JPN}$	0.001	0.011	0.001	0.038	0.013	0.000	0.285	0.007	0.016	0.000	0.029	0.013	0.008	0.000	0.000	1.111	0.039	0.069
USA	0.260	0.402	0.254	0.119	0.413	0.285	0.000	0.205	0.169	1.141	0.835	0.903	0.930	1.135	1.111	0.000	0.738	0.674
Avg.	0.004	0.033	0.003	0.013	0.036	0.007	0.205	0.000	0.005	0.045	0.008	0.009	0.011	0.043	0.039	0.738	0.000	0.010

Table 10 The Great Recession in G7 Countries.  $\hat{\sigma}\text{-statistics}$  for hours in G7 countries.

					V	ES				CD								
	CAN	DEU	FRA	GBR	ITA	JPN	USA	Avg.G7	Avg.OECD	CAN	DEU	FRA	GBR	ITA	JPN	USA	Avg.G7	Avg.OECD
CAN	0.000	0.042	0.312	0.402	0.012	0.006	0.017	0.031	0.263	0.000	0.774	0.136	0.132	0.832	0.013	1.623	0.269	0.402
DEU	0.042	0.000	0.513	0.642	0.036	0.029	0.052	0.106	0.474	0.774	0.000	0.372	0.495	0.599	0.607	1.116	0.334	0.227
$\mathbf{FRA}$	0.312	0.513	0.000	0.009	0.277	0.308	0.238	0.155	0.018	0.136	0.372	0.000	0.011	0.330	0.067	0.892	0.026	0.070
$\operatorname{GBR}$	0.402	0.642	0.009	0.000	0.374	0.406	0.329	0.227	0.024	0.132	0.495	0.011	0.000	0.303	0.069	0.836	0.032	0.094
ITA	0.012	0.036	0.277	0.374	0.000	0.002	0.002	0.019	0.246	0.832	0.599	0.330	0.303	0.000	0.654	0.137	0.176	0.138
$_{\rm JPN}$	0.006	0.029	0.308	0.406	0.002	0.000	0.006	0.026	0.265	0.013	0.607	0.067	0.069	0.654	0.000	1.376	0.166	0.273
USA	0.017	0.052	0.238	0.329	0.002	0.006	0.000	0.010	0.208	1.623	1.116	0.892	0.836	0.137	1.376	0.000	0.622	0.545
Avg.	0.031	0.106	0.155	0.227	0.019	0.026	0.010	0.000	0.135	0.269	0.334	0.026	0.032	0.176	0.166	0.622	0.000	0.018



FIGURE 1: U.S. Economic Growth.



FIGURE 2: U.S. Labor Share.



FIGURE 3: Wedges Paths.







FIGURE 4: The Contribution of Wedges to Worked Hours.



0.95

0.9

0.85 0.8

1.15

1.1

1.05





FIGURE 5: The Contribution of Wedges to Output (Detrended).



FIGURE 6: The Contribution of Wedges to Investment.



FIGURE 7: The Contribution of Efficiency Wedges.



(a) Output per capita, worked hours and investment.



(b) Output per capita and four wedges.



(e) Investment per capita and its components.

(f) Labor share and its components.

FIGURE 8: The Great Recession with VES production function.



(a) Output per capita, worked hours and investment.



(b) Output per capita and four wedges.



(e) Investment per capita and its components.

(f) Labor share and its components.

FIGURE 9: The 1982 Recession with VES production function.





(b) Output per capita and four wedges.



(c) Output per capita and its components.



(d) Worked hours and its components.



(e) Investment per capita and its components.

FIGURE 10: The Great Recession with Cobb-Douglas production function.





(a) Output per capita, worked hours, and investment.



(c) Output per capita and its components.



(b) Output per capita and four wedges.

(d) Worked hours and its components.



(e) Investment per capita and its components.

FIGURE 11: The 1982 Recession with Cobb-Douglas production function. 57

\*\*\*\*